MRS Drone: A Modular Platform for Real-World Deployment of Aerial Multi-Robot Systems

Daniel Hert · Tomas Baca · Pavel Petracek ·

Vit Kratky · Robert Penicka · Vojtech Spurny · Matej Petrlik · Matous Vrba · David Zaitlik · Pavel Stoudek · Viktor Walter · Petr Stepan · Jiri Horyna · Vaclav Pritzl · Martin Sramek · Afzal Ahmad · Giuseppe Silano · Daniel Bonilla Licea · Petr Stibinger · Tiago Nascimento* · Martin Saska

Received: date / Accepted: date

Abstract This paper presents a modular autonomous Unmanned Aerial Vehicle (UAV) platform called the Multi-robot Systems (MRS) Drone that can be used in a large range of indoor and outdoor applications. The MRS Drone features unique modularity with respect to changes in actuators, frames, and sensory configuration. As the name suggests, the platform is specially tailored for deployment within a MRS group. The MRS Drone contributes to the state-of-the-art of UAV platforms by allowing smooth real-world deployment of multiple aerial robots, as well as by outperforming other platforms with its modularity. For real-world multi-robot deployment in various applications, the platform is easy to both assemble and modify. Moreover, it is accompanied by a realistic simulator to enable safe pre-flight testing and a smooth transition to complex real-world experiments. In this manuscript, we present mechanical and electrical designs, software architecture, and technical specifications to build a fully autonomous multi UAV system. Finally, we demonstrate the full capabilities and the unique modularity of the MRS Drone in various real-world applications that required a diverse range of platform configurations.

Keywords UAV platforms · Research and Development · UAV applications

1 Introduction

In recent years, many research groups have developed Unmanned Aerial Vehicle (UAV) platforms with a wide range of capabilities and applications in mind. Still, only

Corresponding author: Tiago Nascimento Faculty of Electrical Engineering, Czech Technical University in Prague, Czech Republic Tel.: +420 22435 7634 E-mail: pereiti1@fel.cvut.cz

a percentage of worldwide UAV research is tested in real-world experiments [54]. At the same time, research groups and startups starting out with UAV research and development can benefit from open-source platforms, as they significantly decrease their initial development costs. Thus, there exists a need for an open-source platform allowing for easy real-world deployment and a high degree of modularity, such as that presented by the MRS Drone. In particular, the modularity aspect allows the platform to be deployed in different applications and has the potential to push the frontiers of UAV research worldwide for the development of new UAV use cases for important industries. Moreover, the MRS Drone was developed to be deployed within a Multi-robot Systems group, which can further increase its application potential.



Fig. 1: MRS UAV platforms used for a diverse range of real-world applications (a)-(c), (g)-(i) and their simulated variants (d)-(f), (j)-(l), respectively. The numerous applications, ranging from aerial indoor inspection to multi-robot wall building, is enabled thanks to the high modularity of the MRS drone.

Although many UAV platforms have been developed [31], there is still the need for a platform that is simultaneously modular, agile, and robust, while also having a sufficient flight time duration to perform complex tasks and the ability to process highly-demanding algorithms. To this end, the MRS group in Prague¹ has, for the past seven years, developed the MRS Drone platforms shown in Fig. 1. Our system has allowed more than 300 bachelor's, master, and Ph.D. students from more than 100 research groups worldwide to perform real-world experiments in indoor and outdoor

¹ http://mrs.felk.cvut.cz

conditions during the MRS summer schools held in 2019, 2020, 2022, and 2023². The modularity of the MRS Drone has been exploited in a large number of industrial applications, such as firefighting and drone hunting, as well as in robotic competitions, including Defense Advanced Research Projects Agency (DARPA) Subterranean Challenge (SubT) Challenge, Mohamed Bin Zayed International Robotics Challenge (MBIZRC) 2017, and MBIZRC 2020. With the MRS Drone platform, research groups and startups are able to build a multi-rotor UAVs equipped with the MRS Drone Software (SW) system [10] on their own, or with the support of the modular DroneBuilder web page³.

3

The remainder of the paper is organized as follows. An overview of the state-ofthe-art UAV systems is presented in the rest of this section. Section 2 presents the mechanical design of the MRS Drone while Section 3 presents the electrical design. In Section 4, we overview the MRS Drone system architecture. Finally, Section 5 presents the diverse applications of our platform, and Section 6 concludes the paper.



Fig. 2: UAV comparison of research platforms between the MRS (CTU) platform [10,36] and other existing platforms, as provided by an independent study from the well-recognized robotic group of U-Zurich [31].

1.1 Contributions beyond the State-of-the-art

Developing a new UAV platform is not an easy task. State-of-the-art research aims to achieve a UAV that has both high flight agility and a high computational and autonomy capability (see Fig. 2). In addition, the large variety of applications demands different sets of sensors, which in turn changes the technical specifications and power demands of each drone. Moreover, some applications require the inclusion of different types of actuators, such as arm manipulators or even fire extinguishers [61].

³ https://dronebuilder.fly4future.com

² http://mrs.felk.cvut.cz/summer-school-2023/

Nowadays, novel concepts of autonomous helicopter platforms continue to develop [90] as different designed platforms are required for new applications that appear every year. The design of new UAVs continues to experience difficulties (e.g. long design periods, high manufacturing costs, and difficulty in performing hardware maintenance). A recent work [33] presented a novel methodology to design lightweight and maintainable frames of micro aerial vehicles by using a configurable design. Many other aspects were also investigated in recent years such as the cost reduction for UAV construction and maintenance [30], multi-robot systems design with commercial benchmark platforms [104], and the experimental hexacopter platform design and construction [77]. Furthermore, several UAV research groups worldwide have proposed open-source platforms. A summary can be found in a recently published paper from Foehn et al. [31] that presents the most prominent UAV platforms from research groups in the field. In particular, the most recent one from D. Scaramuzza's group from U-Zurich [31] has the best balance between onboard computational capability and agility. There are also platforms from ASL (created by R. Siegwart's group at ETH) [75] and FLA (a joint platform from GRASP and U-Zurich) [52] that have relatively high weights and low agility. On the other hand, the relatively small Crazyflie [32, 78, 81] platform does not allow for sufficient onboard computation or sensing. Finally, the platforms from S. Karaman's group from MIT [5] and GRASP (created by V. Kumar's group from UPenn) [47] do not provide open-source software and hardware. Many proposed platforms try to create generaluse UAV hardware for research purposes. Nevertheless, none of the above-mentioned ones have been repeatedly tested in real-world environments in such a wide range of applications as the MRS Drone. Thus, these UAV platforms do not facilitate the desired minimization of the reality gap between the experiments and real-world deployment. Moreover, as shown in Fig. 2, the MRS Drone has the best computing and autonomy capability, which allows for high modularity in diverse applications.

The years of field experience gained by the Multi-robot Systems (MRS) group has provided the foundation for the state-of-the-art research that has resulted in the proposed MRS Drone system, which in turn has yielded dozens of publications by numerous distinct research groups. Our platforms were used in various demanding robotic applications, such as remote sensing, aerial manipulation, aerial pick-andplace tasks, marine surveillance, and autonomous wall building, as described in Section 5. Such uses have demonstrated the high modularity and versatility of our platform. The Hardware (HW) and Software SW system parts of the MRS Drone have allowed researchers worldwide to perform their experiments in real-world conditions, which in turn has allowed us to improve our platform over time. The key features of the MRS Drone include its modular HW construction, open-source SW and pluginbased architecture, and high computational and autonomous capabilities. The MRS software system has been described in our past works [10,36]. Furthermore, the MRS system provides an actively maintained and well-documented implementation on GitHub⁴, including a realistic UAV ROS-gazebo-based simulation, a large variety of sensor setups, diverse UAV control strategies, and a robust multi-modal localization system.

⁴ https://github.com/ctu-mrs/mrs_uav_system

In contrast to our previous work [36], this is the first time the hardware aspect of the co-designed MRS Drone has been detailed with its experimental hardware results, and its applications in numerous different environments expanded upon. Therefore, the contributions of this work are considered as follows:

- 1. We propose a modular UAV platform that can be used in various applications with distinct actuator and sensor configurations while minimizing the effort required for maintenance and substitution of broken parts. This feature is especially appealing due to the fact that real robot experiments have a high probability of collisions, mainly in the initial stages of experimentation with new approaches, and often need maintenance.
- 2. We present an analysis of the results of experiments that aim to test the propulsion systems of our UAVs with respect to (w.r.t.) vibration during flight.
- 3. Our proposed platform facilitates the transition from simulation and simplified laboratory experiments to the deployment of aerial robots in real-world conditions with a minimal sim-to-real gap.
- 4. We expand upon the applications of our modular UAV platforms, including subterranean environments, package delivery, marine environments, human-UAV interaction, and so on.
- 5. Finally, with the goal of enlarging the user community of fully autonomous-UAV, we designed the system with the intent to support the initial steps of researchers and students from different scientific areas where UAV are a necessary tool for the experimental validation of proposed concepts.

2 Mechanical Design and Prototyping

In this section, we present the mechanical design of our proposed platforms. We discuss the frame and propulsion system selection, the design of 3D printed parts to support the mechanical enhancement of the UAV, the UAV sensor selection, multi-robot sensor selection, the communication technologies used (e.g., UltraViolet (UV) cameras), and the actuators used for specific applications.

2.1 Frames and propulsion system consideration

UAV frames are an essential part of the aircraft. The correct choice of frame results in the size of the aircraft and the maximum diameter of the propellers it must use, which in turn affects the resulting payload the UAV will be able to carry and the endurance it will have. The propeller size and motor choice define the maximum thrust and, therefore, the maximum payload of the UAV. The propeller produces thrust in one direction by accelerating air in the opposite direction. In general, to produce a defined amount of thrust, it is more efficient to accelerate a larger mass of air by a smaller amount, than to accelerate a small amount of air by a greater amount. This means that to produce the same amount of thrust, a larger propeller spinning at a slower speed is more efficient than a smaller propeller spinning at a faster speed [35]. In Table 1, we compare the performance of two propellers (one 8-inch and one 9.4-inch) with

	8045 propeller	045 propeller								
	Throttle (%)	Thrust (N)	RPM	Voltage (V)	Current (A)	Power (W)	Efficiency (g/W)			
	50	3.23	5939	9 16.70 2.73		45.59 7.08				
	60	4.29	6902	16.66	4.17	69.47	6.18			
	70	5.42	7648	16.59	6.06	100.54	5.39			
	80	6.63	8499	16.52	8.22	135.79	4.88			
	90	7.81	9076	16.45	10.50	172.73	4.52			
	100	8.91	9722	16.37	13.27	217.23	4.10			
9450 propeller										
	9450 propeller									
	9450 propeller Throttle (%)	Thrust (N)	RPM	Voltage (V)	Current (A)	Power (W)	Efficiency (g/W)			
-	9450 propeller Throttle (%) 50	Thrust (N) 4.01	RPM 5688	Voltage (V) 16.66	Current (A) 3.09	Power (W) 51.48	Efficiency (g/W) 7.79			
-	9450 propeller Throttle (%) 50 60	Thrust (N) 4.01 5.29	RPM 5688 6508	Voltage (V) 16.66 16.61	Current (A) 3.09 4.80	Power (W) 51.48 79.73	Efficiency (g/W) 7.79 6.64			
	9450 propeller Throttle (%) 50 60 70	Thrust (N) 4.01 5.29 6.56	RPM 5688 6508 7213	Voltage (V) 16.66 16.61 16.54	Current (A) 3.09 4.80 6.72	Power (W) 51.48 79.73 111.15	Efficiency (g/W) 7.79 6.64 5.90			
	9450 propeller Throttle (%) 50 60 70 80	Thrust (N) 4.01 5.29 6.56 7.87	RPM 5688 6508 7213 7788	Voltage (V) 16.66 16.61 16.54 16.48	Current (A) 3.09 4.80 6.72 9.05	Power (W) 51.48 79.73 111.15 149.14	Efficiency (g/W) 7.79 6.64 5.90 5.28			
-	9450 propeller Throttle (%) 50 60 70 80 90	Thrust (N) 4.01 5.29 6.56 7.87 9.11	RPM 5688 6508 7213 7788 8608	Voltage (V) 16.66 16.61 16.54 16.48 16.40	Current (A) 3.09 4.80 6.72 9.05 11.48	Power (W) 51.48 79.73 111.15 149.14 188.27	Efficiency (g/W) 7.79 6.64 5.90 5.28 4.84			
-	9450 propeller Throttle (%) 50 60 70 80 90 100	Thrust (N) 4.01 5.29 6.56 7.87 9.11 10.24	RPM 5688 6508 7213 7788 8608 9068	Voltage (V) 16.66 16.61 16.54 16.48 16.40 16.32	Current (A) 3.09 4.80 6.72 9.05 11.48 14.26	Power (W) 51.48 79.73 111.15 149.14 188.27 232.72	Efficiency (g/W) 7.79 6.64 5.90 5.28 4.84 4.40			

Table 1: Motor and propeller tests were done on a static thrust-measuring stand. Both propellers were tested with the same Readytosky 2312 920KV and a MultiStar BLHeli32 51A ESC.

the same motor and Elettronic Speed Controller (ESC). It is evident that the larger propeller is more efficient and produces higher maximum thrust.

The design of UAVs capable of longer flights and able to carry high payloads requires larger propellers. In contrast, the smaller the UAVs are, the easier to operate they are. The small size of UAVs enables the to fly in obstacle-crowded environments. In the end, there is always a compromise between the UAV size and the propulsion efficiency. The standard MRS platforms are based on three basic frame sets and are suitable for most of our research experiments and tests. Some tasks require unusual designs and capabilities that cannot be accommodated by the standard frames. Therefore, custom task-specific frames are also presented.

2.2 MRS Drone used frames

The frames that are used to support the majority of the recent works have similar construction patterns that use four arms positioned between a central structure. In general, the arms are made from plastic or carbon fiber, while the central structure is made from glass Reinforced Epoxy Laminate (REL) or carbon fiber. Carbon fiber composite is preferred as a frame material, as it offers great stiffness while remaining lightweight. Plastic parts are heavier and not as stiff as carbon fiber, but they are also much cheaper. The top or bottom structural boards can be replaced with a custom-made printed circuit board containing the electronics necessary for the functionality of the UAV. Integration of electronic modules, thereby reducing the mass and complexity of the UAV.

The first and smallest basic frame is the DJI F450, shown in Fig. 3, equipped with 2212 KV920 motors and plastic 9.4-inch propellers. Plastic propellers are not as stiff and efficient as carbon fiber composite propellers, but they are more damage-resistant. This makes them suitable for a smaller and cheaper UAV. The platform is



Fig. 3: DJI F450 frame used for swarming research.

powered by a 4S 6750 mAh lithium polymer battery. This combination offers about 0.5 kg of usable payload and flight times between 10-15 minutes, depending on the payload. This platform is the smallest and cheapest out of the three standard frames and is therefore ideal for swarming research, where a large number of UAVs are required. It is also employed in experiments that do not require larger payloads. The 4S 6750 mAh battery has a capacity of 99.9 Wh, which is convenient for air transport, as most airlines limit the size of a battery that can be carried without any special provisions to 100 Wh.

The mid-sized Holybro X500 frame, showcased in Fig. 4, is the preferred option. Its 3510 KV700 motors and 13-inch carbon fiber composite propellers offer exceptional high thrust and propulsion efficiency. The drone can be powered by either one or two 4S 6750 mAh lithium polymer batteries. To increase the payload capacity and flight duration, our X500 has an enhanced propulsion system compared to Holybro's default kit (2216 KV920 motors with 10-inch propellers). The X500 can carry a usable payload of up to 1.5 kg. If a more extended flight time is necessary, an additional battery can be connected in parallel, enabling over 20 minutes of flight time even with a full payload. Additionally, the individual batteries are under 100 Wh, making it easy to transport by plane. Due to its compact design, high payload capacity, and long flight duration, this platform was used in the DARPA SubT Challenge. Moreover, it achieved the best performance in the DARPA SubT Virtual Challenge, where all teams occupying the first six places in the final competition used the MRS X500. In summary, this platform's compact design, high payload capacity, and propulsion efficiency make it an excellent choice for long flights, indoor research, and carrying heavier sensors.

The Tarot T650, displayed in Fig. 4, is the largest frame chosen for the project. It features 4114 KV320 motors, carbon fiber 15-inch propellers, and a 6S 8000 mAh lithium polymer battery. The T650 is the heaviest standard MRS platform with a payload capacity of 2.5 kg. It offers a higher payload and a notably bigger payload volume than the X500, which can be required for some applications. It was used in the MBIZRC 2020 competition, where large payloads and clearances were required for carrying heavy bricks [9], a ball-catching net [95], and bags of water for extinguishing fires [99].



Fig. 4: Holybro X500 frame with upgraded motors.



Fig. 5: Tarot T650 frame.

2.3 Custom platforms for specific applications

In some cases, the use of a basic frame is not ideal. For instance, drones with higher payloads require bigger frames, sometimes being preferable even a non-square frame shape. A specific setup is also useful for physical interaction with the environment, or due to some special end effector, such as a gripper, magnets, or hooks.

To provide an example, the Dronument project⁵ utilizes a high-payload coaxial octocopter for autonomous inspection of historical buildings [44,65]. This platform is meant to carry a regular-size camera with a dual-axis gimbal stabilization, OS0-128 high-resolution Light Detection and Ranging (LiDAR), ultrasonic sensors, and a custom carbon fiber safety cage to protect the platform and its surroundings. Moreover, the platform size must be minimized to allow for flight in confined environments. To maximize the thrust, coaxial propulsion was used. The drone dimensions stay reasonable, while the maximum thrust is increased and, therefore, the maximum payload. However, the coaxial propulsion setup is less efficient, with 20% more power [15] required to generate the same thrust as a standard propulsion dual motor setup. The platform is shown in Fig. 6.

A platform based on similar requirements and principles was used for extinguishing fires on the above-ground floors [84] of buildings, utilizing a unique pneumatic

⁵ http://mrs.felk.cvut.cz/dronument



Fig. 6: UAV platform developed for the Dronument project.

launcher that was able to discharge a fire-suppressing capsule propelled with compressed CO_2 gas. The capsule (containing water and fire suppressant substances) was able to break through a window of a burning building and extinguish a fire inside. The entire frame was built around the design requirement imposed by the long shape and heavy weight of the launcher. Similar to [44, 65], coaxial propulsion was also used for this task in order to minimize the size of the UAV. This platform is shown in Fig. 35.

A prototype drone hunter [86] for protected no-fly zones was based on a T18 frame by Tarot (see Fig. 34). This octocopter platform with 18-inch propellers had up to 10 kg of payload, which was required for Ouster LiDAR, a Reach Real-Time Kinematic (RTK) module, and a special folding net to capture and carry other drones. After the proof of concept with the commercial frame, the team designed a fully custom platform made out of carbon fiber and CNC-machined aluminum.

A specially modified T650 platform is used for experiments involving flight over water surfaces. This UAV is equipped with a waterproof shell and custom carbon fiber floaters. All electronics are located inside the shell and are thus protected from water. The floaters are required for emergency landings on water and for safe retrieval of the UAV. This platform is used to follow and land on a Unmanned Surface Vehicle (USV), as well as to detect objects on the water's surface. The motivation of this research is to address water pollution using drones, and a waterproof blue robotics gripper will be mounted on the platform in the future. The UAV can be seen in Fig. 7.

2.4 Prototyping via 3D printing

The technology of additive manufacturing quickly found its place in the design process. Every single one of our drone platforms has custom parts made by 3D printing. The parts are designed from the start to comply with the constraints of additive manufacturing. Filaments from PLA and PETG are mostly used, although they can be swapped for more resilient materials like ABS, ASA, and PC. The prototyping phase is quick, even by fast prototyping standards, using cheaper filaments and low resolution, while the final print of a polished part takes more time. The parts can be easily replicated and mounted on the platform, even during experimental campaigns outside



Fig. 7: Waterproof T650 platform with carbon fiber floaters.

of the laboratory. Moreover, designing unique frame-sensor combinations and modifying sensory attachments to adapt to the ever-changing requirements is a fast and simple task, even during ongoing or onsite research.

Some parts made by additive manufacturing include for example, UV-light holders, RGB camera mounts, LiDAR mounts, onboard computer covers, and battery holders. During the design, stress simulation, and shape optimization are used to achieve the best strength-to-weight ratio for the part. This is discussed more in Section 2.5.

A crucial component of the UAV is its set of custom legs. They are optimized to be strong enough to support the platform, while simultaneously being as light as possible. In addition, they are intended to be used as modular holders for primarily sensory equipment. For illustration, a Basler camera and LED lights were mounted on 3D printed legs to provide lighting for better image capture [43,76]. In case a different size is needed, the leg's 3D model can be easily modified and extended, even during experimental campaigns. Another important feature of the leg design as motivated by experience from experiments is the safety of weak spots. In case of an emergency landing, a leg should break at a specific point to absorb the impact energy and, thus, protect the core of the platform. The broken leg can then be replaced in a matter of minutes. While testing novel research concepts that are, in principle, unreliable, these kinds of modifications proved to be crucial in practice.

3D-printed dampers and bushings from flexible filaments can be utilized to avert the unwanted vibrations that are ever-present with a multi-rotor UAV. Figure 8 shows the F450 platform equipped with a monocular camera and an IMU for Visual-Inertial Odometry (VIO). The sensors are mounted on a 3D-printed battery case, with the battery serving as an added mass for improving the damping performance. The battery case is placed on a set of dampers printed out of the TPU 30D filament. Such an approach significantly decreases the vibrations present in the IMU data and enables reliable usage of VIO, whose performance is greatly affected by the propellerinduced vibrations [14, 70]. Moreover, a 3D-printed battery holder with 3D-printed bushings can have a variable position based on the appliances attached and a varying center of gravity of the drone. Figure 9 shows a comparison of vibration spectra

11



Fig. 8: F450 UAV platform with 3D printed vibration dampers between the battery holder and the rest of the UAV for visual-inertial odometry applications: (a) 3D view, (b) cross-section with highlighted dampers.

during flight with and without damping, measured using the ICM-42688 IMU. The IMU was mounted on the damped part of the UAV and measured accelerometer data at the rate of 1 kHz. In the case of the F450 platform, the IMU was mounted on the 3D-printed battery case (see Figure 8), and 3D-printed dampers from the TPU 30D filament were used. In the case of the F330 platform, the IMU was attached to the top-mounted battery holder (see Figure 1a), and rubber-based dampers between the battery holder and the UAV frame were utilized. Peaks corresponding to the propeller rotation frequency and its second harmonic frequency are clearly visible, and their amplitudes are significantly reduced by damping.



Fig. 9: Spectra of vibrations measured in the x-axis (pointing to the front of the UAV) of the accelerometer mounted on the F450 and F330 platforms with and without damping. The damped F330 platform was equipped with rubber-based dampers between the top-mounted battery and the rest of the UAV, while the F450 platform was equipped with 3D-printed dampers. In both cases of damping, the IMU was mounted on the battery case, which served as added mass for improving the damping performance.

2.5 Using vase mode for strong lightweight parts

Further optimization regarding weight must be done; one way to do so is to use lighter and stronger materials, such as aluminum or carbon fiber. Unfortunately, these materials are either expensive or hard to manufacture, which makes them unsuitable for



Fig. 10: Cross section of a plane wing printed in vase mode with inner struts formed by incisions.

the iterative design process. When using 3D printing, we can utilize the so-called vase mode, where only the single outer perimeter of the model is printed. If parts are designed correctly, a high strength-to-weight ratio can be achieved. Also, the absence of filament retractions makes the prints look cleaner and, in combination with the absence of the travel behavior, moves faster. The general idea is that thin incisions are extruded to the basic shape, which then forms the vase perimeters. The incisions should be thin enough that the resulting perimeters melt into each other and form a wall, as seen in Fig. 10. This method was originally pioneered by Tom Staton on his YouTube channel⁶.

2.6 Sensory equipment

The mass of a sensor is critical in the selection process, as it dictates the payload capacity of the UAV. LiDARs show a significant change in capabilities as their mass increases. One of the lightest LiDAR, the Garmin LiDAR-Lite, is just a 1D sensor, which is used to measure the UAV's altitude on most MRS platforms. RPLiDAR A3 is heavier, but it provides much more data, being a 2D LiDAR that produces a planar 360° scan of the surrounding environment. The heaviest LiDARs, which is regularly used on the MRS platforms, is the 3D multi-beam LiDARs manufactured by Ouster or Velodyne. These sensors produce a high-resolution 360° scan in multiple planes. The heavier sensors provide much more useful data, but they are also much more expensive, and the UAV platform has to accommodate them with higher payload capacity.

RGB and RGB-D cameras are very attractive, as they are usually very lightweight and produce a lot of useful data. Depth cameras, like the Intel RealSense⁷, are often used for navigation or to detect obstacles in the path of the UAV. These cameras use onboard processing to calculate the depth image from the stereo camera pair, easing the load on the main mission computer. Apart from the stereo pair, Intel RealSense also integrates a regular RGB camera. However, high bandwidth cameras, which use

⁶ https://youtu.be/QJjhMan6T_E

⁷ https://www.intelrealsense.com/depth-camera-d435/

the USB-3 bus, provide strong interference with the signals used by GNSS. Special care has to be taken to shield the GNSS antennas properly.

13

Cheaper cameras often come with rolling shutters, which cause image distortions due to vibrations and fast movements of the UAV. These effects are detrimental to most computer vision algorithms. To counteract them, a camera with a global shutter can be used. The resolution and frame rate of the camera must also be considered. Processing images with higher resolution can yield better results, but it also requires more computational resources and causes larger processing delays, which can be critical to the performance of the entire system.

Other sensors include sonars, which can be used to measure the altitude of the UAVs above water, where other sensors may fail. Many other sensors for specific phenomena can be used for autonomous flight in special conditions and applications; for example, infrared and thermal cameras can be used to detect objects which are colder or hotter than the rest of the background, RGB cameras with a custom UV-pass filter can detect neighboring drones equipped with UV LEDs in UAV teams, TimePix radiation sensors can be used for radioactive objects tracking [7], and gas detectors can sample air quality and deduce toxicity.

Sensors that are used onboard the MRS UAV platforms are as follows:

- Rangefinder: Garmin LiDAR Lite V3, MB1340 MaxBotix ultrasound,
- Planar LiDAR: RPLiDAR A3,
- 3D LiDAR: Ouster OS1 and OS0 series, Velodyne VLP16,
- Cameras: Basler Dart daA1600, Bluefox MLC200w (grayscale or RGB),
- RGB-D cameras: Intel RealSense D435i and D455,
- GNSS: NEO-M8N and RTK Emlid Reach M2,
- Thermal camera: FLIR Lepton, FLIR Boson
- Pixhawk sensors: gyroscopes, barometers, accelerometers (also available as separate sensors with better performance),
- High-rate IMU: ICM-42688,
- UltraViolet Direction And Ranging (UVDAR) system [103].

Every sensor has unique properties and disadvantages and is suited to different conditions and environments. The MRS system can fuse measurements from multiple different sensors to achieve reliable localization and navigation, even in environments where a singular sensor would be insufficient.

In order to successfully deploy multiple UAVs, the MRS platforms must have the ability to fly and cooperate with one another, even in cases where direct radio communication is unreliable or external localization systems are unavailable. In order to achieve this cooperation, the swarm or formation agents must be able to locate each other using their onboard sensory equipment. While computer vision is a cutting-edge technique for mutual relative localization of robots [25,93,94], its current implementations often suffer from degraded performance in general outdoor and indoor conditions, especially when lighting conditions are a factor, as well as high computational complexity when the UAV's payload is limited. To address this challenge, the MRS UAV platforms have integrated a smart sensor known as the UVDAR system [103], which is showcased in Fig. 11 and Fig. 12.



Fig. 11: Pair of DJI F450-based UAV platforms equipped with the UVDAR system. Note, the ultraviolet LEDs and dual cameras on the sides of the UAV body.



Fig. 12: View from onboard ultraviolet-sensitive cameras used in the UVDAR system. Note, the clarity of the onboard LED markers of neighboring UAVs, despite this image being captured at midday in a desert setting.

This open-source⁸ system can be used to perform robust mutual localization. This is done by using ultraviolet LEDs that emit particular optical identification signals, which are detected and decoded using cameras equipped with an optical filter that significantly attenuates most of the image background [98, 101], and thus simplifies the detection of the optical signals on the camera frame. This method has been tested with great success in multiple real-world deployments of swarms and formations of UAVs [2, 27, 39, 58, 66, 102, 103].

Recently, we have improved the identification capacity of the UVDAR by designing special sets of binary signals for the blinking patterns of the ultraviolet LEDs [16, 18]. One of the key properties of these sets of binary sequences is that two arbitrary binary sequences of the same length are considered different if, and only if, their circular Hamming distance is not zero. This property allows the UVDAR system to distinguish different ultraviolet LED blinking patterns without the need for implementing any type of synchronization and thus reduces the latency of the identification process.

Given the length of the binary sequences used for the blinking patterns, our design algorithm proposed in [16,18] creates a set with a maximum number of circularly different binary sequences, satisfying various properties that simplify the detection of

⁸ https://github.com/ctu-mrs/uvdar_core

15



Fig. 13: A block diagram of the integrated distribution board. Thick lines in the diagram represent power connections, while thin lines show data connections.

the LEDs blinking patterns. Among these properties, we have the maximum duration in which a LED can remain turned off. Limiting this time helps the UVDAR system to better track the blinking LEDs on the camera frame when there is relative motion among theUAVs. The larger the binary sequences, the more UAVs can be identified. However, this length is limited by the stability of the frame rate of the UVDAR camera and by the number of different UAVs. See [16, 18] for further details.

If each UAV is assigned only one different blinking sequence, then the UVDAR system can be used only for positional localization of UAVs. However, if more blinking sequences are assigned to each UAV, then the UVDAR system can also be used to estimate their relative heading. The output of the UVDAR localization is a mean relative pose with error distribution, addressing the noise and ambiguity inherent in real-world vision systems.

A modified version of the UVDAR system has been developed to enable optical communication between the UAVs. This modified system is called UVDAR-COM [39] and allows simultaneous communication and localization between the UAVs. Currently, research is being performed to extend its communication range and increase its bit rate, as well as to enable the utilization of the known measurement error covariances for more reliable sensor-based cooperative behavior.

2.7 Additional actuators

The MRS UAV platforms are versatile and can be employed in various indoor and outdoor applications. Some of these applications require specialized actuators that are unique to the particular challenge. For example, in the 2017 MBIZRC competition, electromagnetic grippers were designed to grasp metallic tokens [48, 83], while in MBIZRC 2020, polyurethane foam blocks were used for wall construction [9, 97]. In the fire extinguishing challenge of the same competition, a water cannon and a device that deploys a fire-suppression blanket were utilized, and in another challenge, a system for intercepting a flying target was required [86].

Furthermore, MRS platforms have been equipped with other types of actuators such as manipulators attached to the UAVs for pick-and-place tasks [95], gimbals for camera stabilization in the Dronument project [44, 45], and even a capsule launcher for fire extinguishing in the DOFEC project [84]. Additionally, an Eagle.One⁹ drone [94]

⁹ https://eagle.one/en



Fig. 14: Thrust measurement configurations: (a) lone propeller, (b) ducted fan, and (c) ducted fan with ducting.

was equipped with a net launcher to capture invading drones in aerial no-fly zones (see Section 5 for more details).

2.8 Thrust measurement with different configurations

Regarding future projects that will encompass unusual drone frames, it is necessary to measure the thrust of a propeller in different configurations. The main goal is to determine to what extent additional hardware near or below the propeller affects its thrust performance. Thus, we conducted some experiments with three types of configurations. They are (1) a lone propeller, (2) a ducted fan, and (3) a ducted fan with ducting (see Fig. 14). In theory, the ducted fan configuration should improve turbulent flow leaking near the edge of the propeller. In contrast, the addition of ducting below the propeller should lower maximal thrust, thereby lowering the efficiency of the setup.



Fig. 15: Experimental test rig for measuring the thrust of the ducted fan with additional ducting.

17



Fig. 16: Thrust and motor current curves with various add-ons.

The whole experimental setup consists of our F450 UAV motor configuration (motor, propeller, and motor holder) attached to a load cell through an aluminum extrusion. Said extrusion was affixed to an experimental test rig, forming a single inertial system with all add-ons. The test rig was set high enough to counteract the ground effect, as seen in Fig. 15. All add-ons were printed on a standard FFF 3D printer and allowed for easy modification. In addition, the speed of the motor was incrementally increased. Thrust and the motor current were also measured, and are plotted in the graphs presented in Fig. 16

From the first graph, we can observe several outcomes. In the ducted fan configuration, the thrust is not affected very much. With the added ducting, we lose approximately 20% of thrust, which is comparable with the information we received from external experts. In the current graph (see Fig. 16), there is a visible 5% lower current requirement using the ducted fan configuration, which means this setup is more energy efficient.

In future testing, we will explore a coaxial propeller configuration with the same setup. More concretely, we will explore the distance between propellers, the angle of attack of each propeller, the number of blades, and, finally, the use of compressor propellers instead of traditional ones.

3 Electrical Design and Configuration

The architecture and modularity of the MRS UAV platforms mean that additional electronic modules are required in order to provide additional features, such as low-level interfaces for communication with sensors and actuators or power supplies of different voltages, among others. The MRS group solved this necessity by designing and manufacturing a series of printed circuit distribution boards for the F450, T650, and X500 platforms, which are integrated into the platforms as structural pieces, replacing the top or the bottom board of the frames. This way, the custom board fulfills multiple roles, is a structural member, distributes power, and also integrates addi-



Fig. 17: Distribution board for the F450 shown with an Xbee MRS module (a), and integrated into the F450 platform with an UVDAR controller MRS module (b).

tional electronics. This reduces the mass of the UAV and reduces additional clutter and wiring.

The custom MRS printed circuit board contains two redundant 5 V@3 A buck converters to power the Pixhawk¹⁰, measures the current drawn from the battery and the battery voltage, routes power through high-current capable traces, and distributes throttle signals for all of the motors. The board also comes with standardized expansion slots to connect other low-level boards called MRS modules, which can provide the UAV with additional capabilities. The communication is facilitated by FT4232H: a USB-to-quad serial converter that connects up to four separate UARTs through a single USB 2.0 cable to the main computer. When connected, the device appears as four separate serial ports on the main computer, which makes the development of accompanying software very straightforward. One of the UARTs is used to provide a link between Pixhawk and the main computer using the Mavlink protocol, while the three remaining UARTs connect to the slots for MRS modules. One additional 5 V power supply powers the MRS modules. A functional diagram of the integrated distribution board is shown in Fig. 13.

MRS modules are small daughterboards that can be connected to the main distribution board through a standardized interface. The electrical connection is through a standard 2.0 mm 10-pin header, which provides UART communication to the main computer through the FT4232H, stabilized 5 V rail to power electronics of the MRS module, and direct battery connection for applications that require higher voltage. The MRS module can draw up to 2 A on both the 5 V rail and direct battery rail. The mechanical connection is through two M3 mounting posts, which are installed into the main distribution board. The module is then connected through the header and secured with two M3 bolts to the mounting posts. The standardized electrical and mechanical interface for the MRS modules is shown in Fig. 18.

Up to three MRS modules can be added to the main board to increase the system's capabilities. For example, one module controls the LEDs of the UVDAR system, allowing up to eight high-power LEDs to be individually controlled and their blinking frequencies to be changed based on commands from the high-level computer. Another module has a slot for XBee radios, providing the UAV with a means

¹⁰ The Pixhawk autopilot is an open-hardware and open-software architecture, which is advantageous for research in the field of aerial robotics [51].

19





Fig. 19: Diagram of the system architecture: *Mission & navigation* software supplies the position and heading reference (\mathbf{r}_d , η_d) to a reference tracker. *Reference tracker* creates a smooth and feasible reference χ for the feedback controller. The feedback *Reference controller* produces the desired thrust and angular velocities (T_d , ω_d) for the Pixhawk embedded flight controller. The *State estimator* fuses data from *Onboard sensors* and *Odometry & localization* methods to create an estimate of the UAV translation and rotation (\mathbf{x}, \mathbf{R}) [10].

of low-level communication. An MRS module with an integrated 12 V buck regulator and power switches can be used to control LED strips attached to the UAV, providing additional lighting for cameras. Figure 17 shows a distribution board with an XBee MRS module.

4 MRS UAV System

The presented subsection briefly describes the MRS UAV System, an open-source software package for controlling and deploying multi-rotor aerial vehicles [10]. The system is designed for use in both realistic simulations and real-world scenarios and is capable of executing complex missions in GNSS and GNSS-denied environments. The system includes two feedback control designs, one for precise and aggressive maneuvers and the other for a stable and smooth flight. Control reference generation is provided by the real-time virtual Model Predictive Control (MPC) tracker [6]. The system is modular, allowing users to develop and test their own feedback controllers, reference generators, and estimators. The MRS UAV control and estimation pipelines use rotation matrices and a novel heading-based convention to represent the one free rotational degree of freedom in 3D of a standard multi-rotor aircraft, instead of the Euler/Tait-Bryan angle representation which can cause confusion due to ambiguities and singularities. The MRS UAV system has been used in various real-world

system deployments that have helped shape the system into its current form, as shown in Fig. 19.

The MRS UAV System also provides features for managing and controlling multi-UAVs systems. The Model Predictive Control (MPC) tracker [6] integrates an outdoor mutual collision resolution system. Mutual collision avoidance utilizes wireless communication for exchanging predicted future trajectories of each UAV, which are consequently used to temporarily alter the UAVs' motion. Therefore, even head-on collisions of an arbitrary number of UAVs are automatically avoided without human intervention. However, the users are **discouraged** from relying on the avoidance system as a *basic feature* of the system when designing their multi-UAV control algorithms. Users should implement their collision resolution primarily in their control design and only use the provided avoidance as a failsafe.

The wireless communication within the MRS UAV system is managed by the NimbRo network package¹¹, which provides a multi-master-like communication scheme between multiple independent ROS-controller robots. Using the NimbRo network package, the ROS topics and services of a single UAV can be exposed on other UAVs for supporting distributed control algorithms. The topics and services can also be exposed to a ground control station, which can control a group of multiple UAVs in a centralized manner.

5 Applications for the Aerial Platforms

One important feature of the MRS system is that both MRS SW and HW are smoothly coupled, independent of the real UAV platform used. This enables the MRS SW system to be easily integrated with the robotic platforms independent of the hardware irregularities and modularities that may exist. This easy integration is often needed in general research.

In this section, we present most of the UAV applications and the required platforms performed by our group. Realistic Software-In-The-Loop (SITL) simulations, robotic competitions, research on aerial autonomy in indoor and outdoor environments, and prototyping for industrial applications are among the case studies presented. Finally, we also present a summary of the principal attributes of our UAV platforms used for these applications in Table 2.

5.1 Key features of the MRS Drone

A main advantage of our proposed platforms is the modularity feature. The modularity of our UAVs can be seen not only in their mechanical design but on the electronic boards and in the SW system. Such a modular design allows our platforms to be applicable in a large range of applications and scenarios. Another advantage is the fix-yourself characteristic of our platforms; as an open-source platform, parts can be easily replaced and some can even be printed. This fix-yourself feature is intended to help young researchers to assemble the UAV themselves, thereby boosting research in

¹¹ https://github.com/AIS-Bonn/nimbro_network, Accessed on 2023-03-24

Table 2: Aerial platforms utilizing the MRS software stack in research, academic, and industry projects. *Dimensions* is represented by the length of the main diagonal without propellers. *Parts* denotes the **Publicly** available and Custom-made parts required for construction. *Purpose* denotes the **R**esearch and **Industrial** platforms.

Platform	X500	F450	T650	Dronument	Eagle.One	DOFEC
Flight time (min)	25	15	20	7	10	10
Weight (kg)	2.5	1.7	3.5	5.5	10	7
Dimension (mm)	500	450	650	570	1250	657
Propeller size (in)	13	9.4	15	12	18	15
Battery capacity (Wh)	199.8	99.9	177.6	355.2	355.2	355.2
Rotors count	4	4	4	8	8	8
Parts	Р	Р	Р	С	С	С
Purpose	R/I	R	R	R	Ι	Ι

aerial robotics. A third advantage includes the high computational and autonomous capabilities of our UAVs, as could be seen in the comparison shown by Fig. 2. In contrast, the main disadvantage of our platforms is the low agility during flight. Nevertheless, our current research is aiming to suppress such disadvantages by creating more agile flight control and state estimation algorithms.

Regarding multi-robot systems, our platforms can be easily used in a swarm of UAVs. Furthermore, we use different technologies, such as a multi-spectral communication suite, traditional Wi-Fi, and low-power lightweight units based on RFM69HCW. Even bandwidth-intensive data up to $1 \,\mathrm{MB}\,\mathrm{s}^{-1}$, such as maps [53] or images, are transmitted over a 2.3 GHz Mobilicom MCU-30 Lite unit. These communication technologies enable our MRS UAV platforms to share data between robots and base stations. The cooperation algorithm itself is based on the application and is given to the user to create.

5.2 Gazebo realistic simulations

To facilitate the real-world deployment of novel aerial platforms, we have created a realistic simulation environment built on the open-source simulator Gazebo¹². The simulation stack is available to the community at the MRS GitHub¹³. The package contains the realistic representation of our most used UAV hardware platforms, such as the DJI FlameWheel series (F330, F450, and F550), Tarot 650 sport, or the Holybro X500. It provides a convenient interface for equipping the platforms with various payload configurations, sensory equipment, and actuators.

We employ the Jinja¹⁴ templating engine to assemble the simulated UAVs according to the user's requirements. A base template is provided for each platform (i.e. propellers, motors, arms, and legs) and for individual components that can be attached to the base. The user specifies the platform and the components to be attached, and the robot description for Gazebo is generated dynamically. As a result,

¹² https://gazebosim.org/home

¹³ https://github.com/ctu-mrs/simulation

¹⁴ https://jinja.palletsprojects.com

the environment is capable of multi-UAV simulations, including heterogeneous UAV groups. Fig. 20 shows the X500 platform in two different flight configurations, and Fig. 21 shows a heterogeneous group of UAVs in a simulated urban environment.

The goal is to simulate the entire hardware setup with a high degree of realism, thereby minimizing the difference between simulation and the real world. This way, the transition from the drawing board to real applications can be made seamlessly and with fewer safety risks.



Fig. 20: The modularity of the MRS simulation stack: the X500 platform is equipped with the Ouster LiDAR and the Realsense D435 (a), and a swarm configuration equipped with the UVDAR system including UV LEDs and three Bluefox cameras (b).



Fig. 21: A snapshot from the Gazebo simulator showing a heterogeneous group of UAVs in a simulated urban environment.

5.3 Indoor real robot experiments

Our proposed platforms are intended for both indoor and outdoor applications. In this subsection, we focus on indoor applications, such as historic monument documentation, the Search and Rescue (S&R)-motivated DARPA SubT Challenge, and industrial inspection applications. Due to quick modifications of the platform realized by changing sensors and actuators, we showcase the ability to conduct complex mission objectives in multiple indoor environments. The indoor sensory payload is able to sense the surroundings of the UAV to create an environment model, which is then used for navigation.

5.3.1 DARPA Subterranean Challenge — S&R competition in underground environments

The SubT was organized by DARPA [59] in the years 2017-2021 to push the limits of robot deployment in underground S&R operations. In the competition, the robotic team was sent to search for mannequins representing survivors, their belongings, and other specific objects into a previously unvisited underground space. This space consisted of a combination of three domains: mining tunnels, natural caves, and manmade urban structures. The harshness of each environment type posed demanding challenges on the robot autonomy, which challenged the robustness of each module of the developed solutions. While the complicated topology of the unknown environment, which consisted of both narrow corridors and large caves/rooms, tested the navigation capabilities, the sensory perception was challenged by the severe degradation of sensing conditions. The insufficient illumination combined with airborne particles of various origins (e.g., dust, fog, smoke) required a complementary multimodal perception payload for reliable self-localization, the building of an accurate model of the environment, and the detection of search objects. The requirement of a significant sensory payload competes with the need for minimization of the UAV in order to traverse such constrained corridors, as well as the need for a long flight time, in order to get to the furthest parts of the searched space. Throughout the five-yearlong competition, the platform developed evolved significantly [43, 64, 67, 74], with each iteration improving upon the last.

The platform developed for the final event of DARPA SubT was based on the compact and lightweight Holybro X500 quadrotor frame. The size of the carbon fiber frame could be adjusted to accommodate both 13-inch and 14-inch carbon propellers, which were paired with MN3510 KV700 motors from T-motor driven by easily configurable Turnigy Bl-Heli 32 51A ESCs. This propulsion system provides high efficiency and large payload capacity for mission-critical sensors.

The top board of the X500 frame is replaced by the custom MRS distribution board described in Section 3. The board provides a communication interface between the main Intel NUC i7-10710U computer, Pixhawk flight controller, and two MRS modules. One of the modules controls the 12 V LED strips that help achieve lower exposure times of RGB cameras. The second module is an XBee radio receiver that allows the organizers to issue an emergency land command at the end of the mission or to prevent an accident. Regarding power distribution, the distribution board pro-



Fig. 22: The most important components of the X500 UAV platformed designed for the DARPA SubT Challenge. Blue arrows show the connections between individual modules.

vides 4S Li-Pol battery voltage (14.0 V to 16.8 V), which directly powers the ESCs and the main computer. A total of three 5 V/3 A buck converters are used with two providing a redundant power source for the Pixhawk flight controller, and the remaining for powering the MRS modules. The 3D LiDAR is powered by a 24 V/2 A boost converter. The long flight time of 20 min with all sensors and total mass of 3.3 kg is achieved by connecting two 4S 6750 mA h Li-Pol batteries in parallel. The advantage of using two smaller batteries lies in easier transportation since batteries under 100 W h can be taken in carry-on baggage.

The perception payload consists of a horizontally mounted OS0-128 Ouster LiDAR as the main sensor for establishing a model of the traversed environment and preventing collisions. A wide vertical field of 90° is covered by 128 scanning lines of the sensor, and the blind spots above and below the UAV are covered by two Realsense D435 RGB-D cameras. The full coverage of the UAV's surroundings enables collision-less motion in cluttered environments with high verticalities, such as caves or staircases. The bottom-facing camera can also be used for safe landing spot detection. The two RGB-D cameras, together with two RGB Basler Dart daA1600 cameras on the front legs, provide image streams for the object detection neural network running on the integrated GPU of the Intel NUC.

For multi-robot cooperation, the UAV was equipped with a multi-spectral communication suite. In addition to the traditional Wi-Fi, the UAV can share data over two independent communication channels. A low-power lightweight unit based on a RFM69HCW transceiver provides a low bandwidth of 100 B s^{-1} at 868 MHz or 915 MHz for sharing critical mission data. Bandwidth-intensive data up to 1 MB s^{-1} , such as maps [53] or images, are transmitted over 2.3 GHz Mobilicom MCU-30 Lite unit. Both technologies implement a meshing solution for *ad-hoc* connections of arbitrary combinations of deployed robots.

25



Fig. 23: Verification of the X500 platform in the following environments: dark and humid caves (a,b), man-made urban structures (c,d), and dusty ruins of industrial buildings (e,f).

Figure 22 shows the X500 platform with the most important components. The platform was also replicated as a simulation model in the virtual track of the DARPA SubT, where it was deployed by most of the teams and was also part of the winning solution. In total, our team was composed of five UAVs and two Unmanned Ground Vehicles (UGVs). We finished second in the virtual track, and we achieved sixth place in the competition with physical robots, where three UAVs together with six UGVs found seven artifacts in a complex confined environment. The developed approach is extensively covered by [29, 68]. The platform also proved its reliability in intensive tests in intricate environments, such as vast natural caves, ruins of industrial buildings, underground fortresses, mining tunnels, and large-scale outdoor environments with dense vegetation. Some of the environments where the platform was experimentally verified are shown in Fig. 23.

5.3.2 Dronument: documentation of historical monuments

The documentation of difficult-to-access areas of valuable historic buildings puts special emphasis on the robustness and reliability of the deployed systems. The realiza-



Fig. 24: Deployment of cooperating platforms tailored for the realization of advanced documentation techniques in difficult-to-access areas of historical buildings. Video: https://youtu.be/-_1Fjr58a28

tion of this task in complex environments requires minimizing dimensions, while still providing sufficient payload capacity for all necessary sensory equipment for operation in dark conditions. The MRS hardware stack includes platforms designed specifically for autonomous cooperative high-resolution photography in the interiors of buildings based on more than two years of experiments with the preliminary testing platforms. The developed primary platform is an eight-rotor helicopter with a coaxial arrangement of rotors and a custom frame composed of aluminum profiles connected by carbon-fiber elements, including parts preventing collisions of the propellers with the environment.

The equipment of the platform comprises a powerful computational unit Intel NUCi7 onboard computer and a wide range of sensors to maximize the onboard perception and allow for sensory redundancy. The platform further disposes of a dual-axis gimbal for stabilization of the main documentation sensor, with a weight of up to 700 g — usually, a camera operating in the visible spectrum. However, the high modularity of the system allows for its replacement by, e.g., UV, IR, or a multi-spectral camera. To facilitate the process of autonomous image capturing, the camera and stabilization device are connected to the main distribution board, providing an interface to control the shutter trigger and tilt of the camera from an onboard computer. In order to support a human supervisor with information about proximity to obstacles, the status of failure detection systems, and mission diagnostics, the UAV is equipped with a powerful onboard LED with adjustable colors that encode the health status of the UAV (see Fig. 25(d)).

In several documentation techniques used by experts in the field of restoration and historical sciences, the primary platform must be coupled with a tightly cooperating secondary UAV to allow for positioning light at a predefined angle with respect to the scanned object. Lower requirements on the payload capacity of the secondary platform allow for the use of smaller UAV based on the X500 frame, similar to the platform presented in Section 5.3.1, equipped with a powerful directional light source (see Fig. 24). The developed platforms enable the implementation of advanced documentation techniques in hard-to-reach places in proximity to scanned objects. Thus, they collect valuable data for monitoring the state of historic objects and restoration works planning.

27



Fig. 25: Platform (a) tailored for documentation of structures and valuables within interiors of historical monuments (b-d). Videos: https://youtu.be/-_1Fjr58a28 and https://youtu.be/ wyJdtzas1RI.

The main goal of the Dronument¹⁵ project, whose results are summarized in [14, 44, 45, 63, 65, 76, 82], is the deployment of safety-critical platforms (see Fig. 25) as part of a UAV team for the documentation of historical buildings. The presented platforms have already been deployed in 15 historic objects, including two UNESCO World Heritage sites. During this long-term deployment, the system collected almost 11,000 images in more than 200 flights in historical buildings of various dimensions and characteristics.

5.3.3 Industrial inspection

Indoor inspection tasks have similar requirements on robot capabilities as the documentation of historical monuments. Similarly equipped platforms are typically deployed with onboard LiDAR sensors for obstacle detection, as well as self-localization and high-resolution cameras for inspection. However, in contrast to historical buildings, indoor industrial environments, such as maintenance shafts, manufacturing halls, or storage bays, often consist of repetitive and feature-sparse patterns (both visual and geometrical) with small and difficult-to-detect obstacles (e.g., ropes, cables, poles). A higher resolution sensor onboard the UAV and specialized self-localization and obstacle detection algorithms must be employed to tackle these challenges. Therefore, the UAV platforms require a trade-off between high payload capacity for carrying these sensors and maintaining a sufficiently compact form to pass through narrow corridors. The risk of collisions with environmental obstacles can also be addressed

¹⁵ http://mrs.felk.cvut.cz/dronument



Fig. 26: Snapshots of deployment of industrial platforms (a,b) and a general research platform (c) for inspection of ventilation systems, structural degradation, and electrical infrastructure in tunnels, halls, and storage houses. Video: https://youtu.be/60nKXamV2ds.

with the mechanical protection of propellers and sensors. Examples of the deployment of our custom UAV in these scenarios are showcased in Fig. 26.

5.4 Outdoor real robot experiments

For outdoor experimentation, the MRS platforms possess a versatile sensory system that allows the UAV to vary between full reliance on GNSS data, to deploying a multirobot system in a GNSS-denied area (e.g., forests). In this subsection, we demonstrate the adaptability of the MRS platforms for use in a diverse range of outdoor scenarios and applications, including swarming in plain fields, desert, forest, and grass hill environments, power-line monitoring, human-robot interaction, and marine applications. In order for the robotic platforms to be used in such expansive environments, modifications on the frame must be performed. Such modifications can include different landing gears, water-proof encapsulation, sensor-fusions techniques, communication shielding, and so on.



Fig. 27: Swarms of UAVs in the desert (a) using MRS platform and 3D formation of 10 UAVs (b). Video: https://youtu.be/o8bphtbPCaA.

5.4.1 Swarming in deserts, hills, and forests

The swarm is a multi-robot configuration in which robots are virtually merged to form a compact group (see Fig. 27) of dispensable and interchangeable robots [73, 105].

29



Fig. 28: Swarms of UAVs in grass hill (a, b) and forest environments (c). Video: https://youtu.be/ HH78AheC-DM.



Fig. 29: Swarm experiment in a desert environment using the optical flow method with a monocular camera for self-localization and supported by a novel multi-robot state estimator. Video: http://mrs.felk.cvut.cz/iros-2022-estimation.

Biological swarms take advantage of a high number of group members to travel long distances and protect individuals from predators. In aerial robotics, swarms of UAVs increase their efficiency, scalability, and reliability in applications, such as environmental exploration and monitoring or search and rescue operations. Nevertheless, bringing these swarm systems into the real world is a long-standing problem, due to the significant operative differences between laboratory environments and diverse, real-world environments. To deal with the demands of various real-world environments, the MRS platforms can be modified to handle the operating conditions required in plain fields, forests, dunes and deserts, hills, and much more [2, 4, 23, 27, 37, 58, 66, 71]. One essential element of biologically-inspired swarming architectures is the perception-aware system, which allows for maintaining group cohesion and avoiding collisions between teammates. Our technological solution to this problem lies in using a communication-less mutual localization system called UVDAR.

The swarm systems introduced in this section are fully decentralized, and thus the behavior of individual UAV is influenced only by the processing of onboard sensory information. Using a 2D LiDAR, a swarm of UAVs was able to navigate through a dense forest in [2, 27, 46]. The swarming solution in [2] used a bio-inspired control approach, where a new approach for avoiding collisions with agents and obstacles to emphasize the safety of the UAV was introduced. In [46], shared obstacle maps were used to perceive neighboring agents. This method showed better results than commonly used vision-based perception systems since the method from [46] is not

influenced by the occlusion of obstacles. A similar flocking model without any communication was introduced in [66]. Moreover, we provided an analysis of the stability of the swarm concerning the accuracy of position estimation. A compact group of MRS UAVs was deployed in a desert environment [37] for search and rescue applications with a CNN camera detector for human-victim detection in a control loop. These experiments also verified MRS platforms under demanding light and temperature conditions. A similar experimental verification was performed on a grassy hill, as shown in Fig. 28, where the swarm system demonstrated the ability to follow over uneven terrain. In [58], a bio-inspired evasion approach in a self-organized swarm of UAVs was introduced. Using the UVDAR system, MRS UAVs were able to avoid dynamic objects (predators) that were actively approaching the group. In the same line, [1] proposes an approach for achieving decentralized collective navigation of UAV swarms without communication and without global localization infrastructure. The work addresses several challenges related to the real-world deployment of a swarm of UAVs and their collective motion in a cluttered environment. Compared to previous works [2, 27], the approach uses relative localization information, recorded over a finite time horizon, to overcome the challenges of occlusion of UAVs by obstacles. In [38], a decentralized multi-robot state estimation approach was introduced to support onboard state estimation performance by swarm aggregation. Using this unique technique, MRS UAV platforms were able to fly above the sand's surface using a visual type of odometry (see Fig. 29), which proved to be unreliable over the uniform surface in standalone mode.

5.4.2 Marine environment applications

Another application tackled by the MRS system is the employment of UAVs in a marine environment [19, 34, 56]. Our UAVs were modified by integrating the principles and theoretical backgrounds of the behaviors of a UAV and/or a swarm of UAVs. New control, localization, and perception approaches were employed in a single robot and in a multi-robot system to enable them to be adapted to an unfriendly environment, especially in cases like UAVs landing on water platforms. Thus, we were able to successfully apply the MRS system in the realistic marine scenarios of surveillance, reconnaissance, search and rescue, and so on. This success was achieved through the development of new techniques that enable the UAVs to interact with USVs and objects on water and to interact with the water environment from an aerial perspective.

Among the applications investigated by our group are the physical interaction of a UAV with a USV, object detection on water, and the picking of debris on the water's surface by UAVs (see Fig. 30). Furthermore, we developed a new UAV platform that contains a new set of sensors with the proper mechanical adaptation needed for such applications in the real world. This new platform was based on the Tarot T650 frame (see Fig. 7) and contained floating gear to ensure safety when landing on water. In addition, all sensory equipment was optimized to enable this UAV to autonomously fly over marine surfaces, such as rivers, lakes, and seas, within any weather conditions (e.g., rain, wind, waves).



Fig. 30: Landing on a USV (a), detecting objects on water (b), and pulling debris out of water. Video: http://mrs.felk.cvut.cz/ral-landing-on-usv.



Fig. 31: Human operator performing actions in a sunny outdoor setting (top) and low-light outdoor setting (bottom).

5.4.3 Human-Robot Interaction

Human-Robot interaction (HRI) systems aim to ease the interaction between humans and robots. This is a growing area of research with open problems, especially when subjected to real-world conditions. The HRI using UAVs makes the HRI problem even harder. Control systems applied to HRI are also difficult to design due to the fact that they must be intuitive for the user. Recently, UAV swarms are being used in HRI problems. One recently published work [40] proposed a giant virtual architecture to enable the operator to control a swarm of robots in different ways, preserving the system's autonomy while enabling overall control of the mission objective. Another method found in the state-of-the-art of HRI techniques is to control a robot leader within a swarm as an avatar of the operator. The robot swarm would then follow the robot leader operated by a remote user [49].

Recently, the MRS group has applied the MRS system to the HRI problem using a third method, in which the human operator controls a swarm of UAVs using visual actions. Although gesture recognition is not new [3, 72], the application of gesture recognition by the onboard cameras of multiple UAVs with limited computational resources continues to be an open problem. The MRS UAV platforms can perform such action recognition and use it for the direct control of a UAV swarm (see Fig. 31).

5.4.4 MBIZRC 2017 and 2020



Fig. 32: MBIZRC 2017: autonomous landing on a moving vehicle (a) and cooperative collection of objects (b); & MBIZRC 2020: autonomous capture of agile objects (c), cooperative wall building (d), and autonomous fire-fighting (e). Video: https://youtu.be/DEUZ77Vk2zE.

The MBIZRC 2017¹⁶ and 2020¹⁷ competitions were composed of various robotic challenges motivated by the intent to push technological and application boundaries

¹⁶ http://mrs.felk.cvut.cz/mbzirc

¹⁷ http://mrs.felk.cvut.cz/mbzirc2020

beyond the current state-of-the-art (see Fig. 32). These technological challenges included fast autonomous navigation in semi-unstructured, complex, and dynamic environments with minimal prior knowledge, robust perception and tracking of dynamic objects in 3D, sensing and avoiding obstacles, GNSS-denied navigation in indooroutdoor environments, physical interaction, complex mobile manipulations, and airsurface collaboration.

33

In MBIZRC 2017, the MRS platform participated in two challenges. Challenge 1 featured an autonomous landing on a moving vehicle (see Fig. 32(a)). The task was to autonomously locate a vehicle, match its speed, and finally land on its roof, all of which used exclusively onboard sensors, such as a camera to detect its pattern and GPS for localization. The MRS system achieved 2nd place in the Challenge 1¹⁸ [11, 87]. The second challenge in MBIZRC 2017 that the MRS platform took part in was Challenge 3¹⁸. This challenge featured a team of three UAVs tasked with finding a set of static and moving colored ferromagnetic objects (see Fig. 32(b)) with consequent delivery of the objects to the target location. Therefore, the UAVs had to autonomously detect the objects with onboard cameras, be able to grasp the object with an electromagnetic gripper, and also coordinate with its multi-robot team to prevent collisions among the UAVs. The MRS platforms won Challenge 3 [48, 83].

In MBIZRC 2020, the MRS platform participated in all three challenges. The task of Challenge 1 was to autonomously track a flying object (enemy drone) and interact with it in order to catch a ball attached to the object (see Fig. 32(c)). The MRS platforms took 2nd place in this challenge¹⁹ [85,86,95]. In Challenge 2, the task was to autonomously build a wall using a team of three UAVs²⁰ and one ground vehicle [9, 97]. The employed MRS platforms were able to place the largest number of bricks on the wall (see Fig. 32(d)), and therefore achieved 1st place in the challenge. Next, Challenge 3 featured a fire-fighting task²¹ as shown in Fig. 32(e) [41,69,84,89,99, 100]. Finally, the Grand Challenge combined all three challenges and was dominated by the MRS UAVs, which collected the highest score among all participants to secure first place.

5.4.5 Aerial-Core

This section describes the results obtained within the H2020 European AERIAL-CORE²² project. The goal of the project is to develop cognitive aerial platforms for the application of autonomous power line inspection and maintenance. This scenario is a typical example of UAV deployment in unknown and safety-critical environments where obstacles (e.g., branches, vegetation, balloons) may affect the autonomous navigation of the vehicles, and therefore the mission accomplishment. Two tasks of interest are considered: (i) *inspection*, where a fleet of UAVs perform a detailed investigation of power equipment by assisting a human operator in acquiring views of the power line (e.g., tower, cables, and insulators) that are not easily accessible, while

¹⁸ https://youtu.be/ogmQSjkqqp0

¹⁹ https://youtu.be/2-cLSjRCKDg

²⁰ https://youtu.be/1-aRtSarYz4

²¹ https://youtu.be/08QBiAyP2c0

²² https://aerial-core.eu



Fig. 33: Snapshots of the MRS drones performing *inspection* and *monitoring* operations in the AERIAL-CORE European project. A power line mock-up scenario is used to show the viability of the approach. Figures (a) and (b) show the *inspection* scenario, while Fig. (c) depicts the *monitoring* operations. Solid circles are used to indicate the UAVs approaching the tower. Video: https://youtu.be/ZqUCG7xqiE4.

also looking for damage to the mechanical structure and failures of electrical components, as depicted in Fig. 33(a) and Fig. 33(b); (ii) *monitoring*, where a formation of UAVs provides a view of the humans working on the power tower to their supervising team in order to monitor their status and ensure safety, as shown in Fig. 33(c).

In both inspection and monitoring tasks, vision sensors are essential for visual examination and monitoring operations. In the UAV configuration, cameras are mounted in an *eye-in-hand* configuration, i.e., rigidly attached to the body frame of the aircraft. Cameras are also used to mutually localize the UAVs in the surrounding environment using the UVDAR system. The so-designed platform is suitable for both indoor and outdoor use, with an emphasis on onboard multi-sensor fusion. This allows vehicles to precisely detect their surroundings and flawlessly perform safe maneuvers in confined spaces.

The UAVs dedicated to the inspection task operate in a known environment represented by a previously acquired map, including the position of the power towers, cables, and vegetation. Additionally, precise algorithms are used for UAV localization and navigation [10]. The onboard controller encodes the mission execution as a task allocation and trajectory planning problem, where the vehicles need to move from an initial position through a sequence of target regions (places that need to be inspected) while avoiding obstacles and maintaining a safe distance between them [20, 79]. The sequence of target regions is assigned to the UAVs by solving a Vehicle Routing Problem (VRP) so that all targets are visited by the UAVs just once while minimizing the mission time. At the same time, the UAVs dealing with the monitoring operations can



Fig. 34: Development of the *Eagle.One* autonomous aerial interception system based on the MRS hardware platform: (a) the first prototype using the MRS UAV platform during a successful interception; (b) the second prototype was designed by the MRS team specifically for the *Eagle.One* project. Video: https://youtu.be/hEDGE7ofX1c.

detect the human gestures [60] used to provide high-level actions, such as requests for new tasks or new parameters for a previously requested task. The onboard controller solves a leader-follower formation control problem where the vehicles need to keep the human worker within the camera field-of-view during the entire operation while providing complementary views from multiple UAVs [42, 62]. Safety is ensured by computing safe corridors with integrated information from a local map and human tracking trajectories. In both inspection and monitoring tasks, the UAVs support maintenance operations of critical power grid infrastructure. Further details about the tasks, the designed algorithms, and the software architecture can be found in [17, 20–22, 26, 28, 42, 55, 57, 79, 80]. Illustrative videos of the experiments using the MRS UAV platforms are available²³.

5.5 Industrial collaboration

The MRS hardware and software platforms have kick-started several industrial projects, serving as a basis for preliminary proof-of-concept prototypes and tests. These collaborations were followed by further development of specialized platforms.

5.5.1 Airspace protection

The *Eagle.One* autonomous aerial interception system is co-developed by our team as part of an industrial collaboration [93–96]. Several prototype platforms were designed as part of this project for evaluating the developed detection and interception algorithms (see Fig. 34). The first prototype served only to test the trajectory planning for following and capturing a target using an onboard net launcher. It was based on

²³ https://mrs.felk.cvut.cz/projects/aerial-core

our modular UAV platform utilizing the F550 frame and was equipped with standard onboard sensors, the PixHawk flight control unit (FCU), the NUC computer, an RTK-GPS for accurate positioning, a camera for detection of a visual marker placed on the target, and the net launcher. The second iteration was designed with the capabilities of detecting a non-marked target and carrying the captured target in mind. For this purpose, a specialized prototype platform based on the Tarot T18 with a payload capacity of 12 kg and a stereo camera for detection was developed. Based on the results of these experimental tests, the "detection and catching" approach was reworked to rely on a 3D LiDAR sensor and a hanging net instead of the net launcher. This enables the interceptor to retry failed catching attempts and provides a 360° field-of-view, which improves the re-detection of the target in such situations. The final prototype is a custom-made UAV octocopter platform with a 3D LiDAR sensor, a deployable net suspended below the interceptor, and a lightweight carbon frame to improve the carrying capacity and leave a sufficient thrust margin for dynamic maneuvering. This prototyping approach enabled us to iteratively improve both the algorithms and hardware, and to effectively converge to a functioning system capable of solving the complex task of autonomous aerial interception.

5.5.2 Fire extinguishing

Multi-rotor UAVs are capable of quickly reaching great heights while carrying liquids, or extinguishants, that are able to extinguish fires located inside buildings. When evaporated inside a fire, an extinguishant effectively pushes oxygen out of it, thereby removing its fuel and consequently extinguishing the fire. Furthermore, exteriorflying UAVs carrying capsules filled with such a fire extinguishant are able to operate near multi-floor buildings. In order to transfer the extinguishant from the UAVs and to the fire, the liquid-filled capsules are launched from onboard the UAVs.

The UAV platforms and mechanisms for launching the capsules, together denoted as DOFEC²⁴, are custom made. There are two DOFEC versions with distinct dimensions and payload capacities. Both versions are built on octocopter UAV platforms carrying interoceptive and exteroceptive sensors (IMU, on-board computer, RGB-D, thermal camera, Ouster LiDAR, and downward rangefinder) required for UAV localization, UAV control, fire detection, fire localization, and positioning near high-rise buildings. The UAV is localized on the basis of the inertial measurements, the GNSS, and the laser-based SLAM running onboard. The fires are localized by fusing onboard thermal and RGB-D cameras.

The small and large platforms are shown in Fig. 35. The smaller prototype carries a single 500 ml capsule which gives the UAV only one chance of launching it into the fire. The larger prototype carries a launcher mounted with six of these capsules, allowing for longer operation, wider room for error, and the capability of extinguishing larger fires. The larger platform has rectangular dimensions of 119×83 cm between motors, 30-inch propellers, weighs 41 kg when fully loaded with six capsules, and reaches a maximal flight time of 15 min. Both the single and six-capsule launchers utilize pressurized CO₂ to fire the capsules. With autonomous UAV control from

²⁴ http://mrs.felk.cvut.cz/projects/dofec

the MRS UAV system, both platforms seamlessly handle the dynamic recoil from launching the capsules.

37



Fig. 35: DOFEC platforms (small platform left column, large platform right column) capable of autonomously launching capsules filled with a fire extinguishant into thermal sources. The top row shows CAD models of the platforms, whereas the bottom row shows them during the mission. Video: https://youtu.be/QHpifXJzH5g.

5.5.3 Localization of radiation sources

An autonomous UAV for nuclear site surveying and homeland security applications was developed through our partnership with the RaDron²⁵ project. The project aims to develop an aerial platform equipped with compact radiation detectors for autonomous operations in hazardous environments. Over the course of the project, we have developed several platforms of varying sizes and payload capacities. The final version of the platform is derived from the Holybro X500 in a configuration similar to the one used in the DARPA SubT Challenge (described in Section 5.3.1). The qualities proven in the challenge translate well in environments where radiological surveys are to be conducted, such as mine shafts, forests, and industrial objects.

The sensory payload is adapted for both indoor and outdoor operations. In GNSSenabled environments, the vehicle relies on the NEO-M8N receiver for positioning. However, it also carries an Ouster OS0-128 LiDAR for obstacle detection and GNSSdenied localization.

The platform is also equipped with a miniaturized Compton camera, which enables the measurement of radiation intensity, as well as the reconstruction of possible directions toward its source. The Compton camera is based on the cutting-edge

²⁵ http://mrs.felk.cvut.cz/projects/tacr-radron-project



Fig. 36: Miniature Compton Camera MiniPIX Timepix3 (a) used as an onboard radiation sensor for ionizing radiation localization by UAVs (b,c). Video: https://youtu.be/oH4jMMhfGVA.

Timepix3 chip sensor developed by the Medipix3 collaboration²⁶. The sensor is part of the pen-sized MiniPIX Timepix3 USB device (see Fig. 36(a)). Contrary to similar state-of-the-art devices, the MiniPIX Timepix3 Compton camera only uses a single detector [92]. This dramatically reduces the size of the device and removes the need for active cooling.

In our applications, we attach the detector to the UAV as a forward-facing camera [8,88]. This configuration enables operation much closer to the ground, and therefore closer to radiation sources than the downward-facing camera used in related projects [24, 50, 91]. The camera is protected by a 3D-printed casing with a small passive aluminum and graphene heat sink at the top, as shown in Fig. 36.

To facilitate the use of Timepix-based devices in other robotic systems, we have developed a ROS interface Rospix [13] and released it as open source²⁷. Rospix enables real-time data readout and is a convenient way to adjust measurement parameters on the fly. Together with the device's USB interface, the MiniPIX Timepix3 Compton camera may be used as a plug-and-play device. The aerial platform is capable of systematic radiation mapping, aerial surveillance, and fast proactive localization of compact radiation sources [8, 12, 88]. Videos of the platform in action are available at the RaDron project web site²⁸.

5.5.4 Asteroid inspection

Asteroids, moons, and other planetary bodies are a rich source of information about our solar system and the origin of life itself. However, such complicated terrain makes it challenging to use ground robots in these environments. We have developed a platform equipped with Ouster LiDARs to perform autonomous mapping and navigation in hard-to-navigate environments (see Fig. 37), which can potentially carry several other sensors for collecting information about the planetary body. Our partners²⁹ have used the platform to evaluate their algorithms for targeted asteroid landing, gravitational field exploration, and topography and resource localization.

²⁶ https://medipix.web.cern.ch/medipix3

²⁷ https://github.com/rospix/rospix

²⁸ http://mrs.felk.cvut.cz/projects/tacr-radron-project

²⁹ https://www.unibw.de/lrt9/lrt-9.1/forschung/projekte/kanaria_nakora

39



Fig. 37: Prototype platform used for exploration, mapping, and data collection for the asteroid inspection project.



Fig. 38: Platform used for autonomous delivery of packages from a warehouse. Video: https://youtu.be/C0p514rz14c.

5.5.5 Package delivery

Robots are widely used for logistical applications and have been utilized in several industries, including warehouse automation and assembly lines in manufacturing. Al-though autonomous UAVs are difficult to use in indoor environments, they have been extremely useful for package delivery applications. Advancing in this direction, we show in Fig. 38 a UAV platform developed by our group with a manipulator for quick package delivery³⁰. Although the UAV was used to deliver commercial packages, it can be used for delivering medical and other critical supplies as well. The quick response time of UAV-based delivery offers a significant advantage over conventional transportation methods.

5.5.6 Indoor inspection

Similar to the inspection of outdoor environments like buildings, power lines, and construction sites, UAVs can be very beneficial for inspecting large indoor spaces. Equipped with Velodyne LiDAR's Puck LITE-16 and several other optical sensors, we developed a UAV for autonomous warehouse inspection. The UAV used a path planning method to autonomously map the environment and avoid obstacles, such as warehouse shelves and cargo. We used an onboard 3D camera to capture high-quality pictures and depth information for later processing to look for specific regions in the

³⁰ https://youtu.be/C0p514rz14c



Fig. 39: Autonomous inspection of a warehouse and an abandoned building using our UAV platform.

warehouse. We also developed another version of the platform with an Ouster LiDAR and different sensors to inspect old infrastructure for safety and regulatory purposes. Figures 39(a) and 39(b) show the platforms inside a warehouse and an abandoned building, respectively.

6 Conclusions

In this work, we have presented the MRS UAV hardware and software system comprising our MRS Drone, which has been proposed to enable research groups to perform real robot experiments with UAVs in real-world conditions. The proposed platform is modular and the software system is open-source, allowing researchers to achieve proper experimental UAV deployment for both a single UAV or multi-UAV system, in either indoor or outdoor scenarios. Furthermore, we have detailed the technical specifications of the UAV platforms, enabling researchers to prototype and design UAVs for research experimentation and validation. Finally, the MRS group has accumulated thousands of hours of experimentation and UAV deployments in a large variety of applications, which were also described in this paper. Therefore, we intend for this paper to facilitate UAV prototyping for autonomous aerial applications in a wider range of scenarios than those presented here.

Declarations

Acknowledgments The authors would like to thank the researchers from the Laboratory of Systems Engineering and Robotics Group at the Universidade Federal da Paraíba that, in partnership, helped improve and test our system. Furthermore, the paper is an extension of the manuscript with DOI: 10.1109/ICUAS54217.2022.9836083, from ICUAS 2022.

Authors' contributions In this work, M. Saska coordinated the entire project and group activities. T. Nascimento (the corresponding author) was responsible for coordinating the writing of this manuscript, and the writing of Sections 1, 5.3.2, and

41

5.3.2. D. Hert coordinated the hardware team and selected the electronic components, actuators, and sensors of the UAVs, and wrote Section 3. D. Zaitlik designed the electronic MRS board. P. Stoudek was responsible for designing the various types of mechanical frames and writing Sections 2.1, 2.2, 2.3, and 2.8, while M. Sramek was responsible for printing all the 3D-printed parts and coordinating the mass production of MRS UAVs and writing Sections 2.4, 2.5, and 2.7. On the MRS system, T. Baca coordinated the MRS software development and wrote Sections 4 and 5.3.4. V. Walter designed the UVDAR MRS system. D. Bonilla Licea helped improve communication through UVDAR and wrote Section 2.6. P. Stepan helped develop the vision system of MRS UAVs. V. Spurny was responsible for developing the simulation environment and wrote Section 5.1. On upgrading the MRS Drone and MRS SW system to attend to various applications, M. Petrlik designed the MRS Drone localization system and wrote Section 5.2.1, while V. Pritzl helped improve the sensor fusion of the MRS Drone and helped write Sections 5.2.1 and 5.4.2. P. Petracek designed the MRS Drone (HW and SW) for the Dronument project and wrote Sections 5.2.2 and 5.4.2, while V. Kratky designed the MRS Drone designed the MRS Drone (HW and SW) for industrial inspections and wrote Section 5.2.3. J. Horyna focused on swarming in deserts and modified the MRS Drone for such applications, as well as wrote Section 5.3.1. M. Vrba focused on the vision system, localization, and AI-based applications of the MRS Drone, while also writing Sections 5.4, 5.4.1, and 5.4.6. P. Stibinger helped improve the MRS Drone SW system and wrote Section 5.4.3. A. Ahmad applied the MRS Drone to outer space applications and manipulation and wrote Sections 5.4.4 and 5.4.5. R. Penicka designed the trajectory and path planning of the MRS Drone, revised the entire manuscript, helped write Section 1, and wrote the abstract and Section 6. Finally, G. Silano wrote Section 5.3.5 and helped revise the entire manuscript.

Code or data availability The authors declare that the data supporting the findings of this study are available within the MRS software link (https://github. com/ctu-mrs/mrs_uav_system) and MRS Drone Builder link (https:// dronebuilder.fly4future.com/).

Consent for Publication Informed consent was obtained from all the co-authors of this publication.

Competing interests The authors declare that they have no conflict of interest.

Ethics approval and consent to participate All applicable institutional and national guidelines were followed.

Funding This work was partially funded by the CTU grant no. SGS20/174/OHK3/3T/13, by the Czech Science Foundation (GAČR) under research project no. 20-10280S, no. 20-29531S, no. 22-24425S and no. 23-06162M, by TAČR project no. FW01010317, by the OP VVV funded project CZ.02.1.01/0.0/0.0/16 019/0000765 "Research Center for Informatics", by the NAKI II project no. DG18P02OVV069, by the European Union's Horizon 2020 research and innovation program AERIAL-CORE under

grant agreement no. 871479, by the Defense Advanced Research Projects Agency (DARPA), and by the Technology Innovation Institute - Sole Proprietorship LLC, UAE. Furthermore, computational resources were supplied by the project "e-Infrastruktura CZ" (e-INFRA LM2018140) provided within the program Projects of Large Research, Development and Innovations Infrastructures.

References

- Ahmad, A., Bonilla Licea, D., Silano, G., Baca, T., Saska, M.: PACNav: A Collective Navigation Approach for UAV Swarms Deprived of Communication and External Localization. Bioinspiration & Biomimetics 17, 6 (2022). DOI 10.1088/1748-3190/ac98e6. URL https://iopscience. iop.org/article/10.1088/1748-3190/ac98e6
- Ahmad, A., Walter, V., Petracek, P., Petrlik, M., Baca, T., Zaitlik, D., Saska, M.: Autonomous Aerial Swarming in GNSS-denied Environments with High Obstacle Density. In: 2021 IEEE International Conference on Robotics and Automation (ICRA), pp. 570–576 (2021). DOI 10.1109/ICRA48506. 2021.9561284. URL https://ieeexplore.ieee.org/document/9561284
- Ahmed, S., Kallu, K.D., Ahmed, S., Cho, S.H.: Hand Gestures Recognition Using Radar Sensors for Human-Computer-Interaction: A Review. Remote Sensing 13, 527 (2021). DOI 10.3390/rs13030527. URL https://www.mdpi.com/2072-4292/13/3/527
- Amorim, T., Nascimento, T., Petracek, P., de Masi, G., Ferrante, E., Saska, M.: Self-Organized UAV Flocking Based on Proximal Control. In: 2021 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1374–1382 (2021). DOI 10.1109/ICUAS51884.2021.9476847. URL https://ieeexplore.ieee.org/document/9476847
- Antonini, A., Guerra, W., Murali, V., Sayre-McCord, T., Karaman, S.: The blackbird uav dataset. The International Journal of Robotics Research 39(10-11), 1346–1364 (2020). DOI 10.1177/0278364920908331. URL https://journals.sagepub.com/doi/abs/10. 1177/0278364920908331?journalCode=ijra
- Baca, T., Hert, D., Loianno, G., Saska, M., Kumar, V.: Model Predictive Trajectory Tracking and Collision Avoidance for Reliable Outdoor Deployment of Unmanned Aerial Vehicles. In: IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pp. 1–8 (2018). DOI 10.1109/ IROS.2018.8594266. URL https://ieeexplore.ieee.org/document/8594266
- 7. Baca, T., Jilek, M., Vertat, I., Urban, M., Nentvich, O., Filgas, R., Granja, C., Inneman, A., Daniel, V.: Timepix in LEO Orbit onboard the VZLUSAT-1 Nanosatellite: 1-year of Space Radiation Dosimetry Measurements. Journal of Instrumentation 13(11), C11010 (2018). DOI 10.1088/1748-0221/13/11/ C11010. URL https://iopscience.iop.org/article/10.1088/1748-0221/13/ 11/C11010
- Baca, T., Jilek, M., et al.: Timepix Radiation Detector for Autonomous Radiation Localization and Mapping by Micro Unmanned Vehicles. In: 2019 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pp. 1129–1136 (2019). DOI 10.1109/IROS40897.2019.8968514. URL https://ieeexplore.ieee.org/document/8968514
- Baca, T., Penicka, R., Stepan, P., Petrlik, M., Spurny, V., Hert, D., Saska, M.: Autonomous Cooperative Wall Building by a Team of Unmanned Aerial Vehicles in the MBZIRC 2020 Competition (2020). DOI 10.48550/arXiv.2012.05946. URL https://arxiv.org/abs/2012.05946. Submitted To Robotics and Autonomous Systems
- Baca, T., Petrlik, M., Vrba, M., Spurny, V., Penicka, R., Hert, D., Saska, M.: The MRS UAV System: Pushing the Frontiers of Reproducible Research, Real-world Deployment, and Education with Autonomous Unmanned Aerial Vehicles. Journal of Intelligent & Robotic Systems 102(26), 1–28 (2021). DOI 10.1007/s10846-021-01383-5. URL https://link.springer.com/ article/10.1007/s10846-021-01383-5
- Baca, T., Stepan, P., Spurny, V., Hert, D., Penicka, R., Saska, M., Thomas, J., Loianno, G., Kumar, V.: Autonomous Landing on a Moving Vehicle with an Unmanned Aerial Vehicle. Journal of Field Robotics 36, 874–891 (2019). DOI 10.1002/rob.21858. URL https://onlinelibrary.wiley.com/doi/abs/10.1002/rob.21858

 Baca, T., Stibinger, P., Doubravova, D., Turecek, D., Solc, J., Rusnak, J., Saska, M., Jakubek, J.: Gamma Radiation Source Localization for Micro Aerial Vehicles with a Miniature Single-Detector Compton Event Camera. In: 2021 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 338–346 (2021). DOI 10.1109/ICUAS51884.2021.9476766. URL https: //ieeexplore.ieee.org/document/9476766

- Baca, T., Turecek, D., McEntaffer, R., Filgas, R.: Rospix: modular software tool for automated data acquisitions of Timepix detectors on Robot Operating System. Journal of Instrumentation 13(11), C11008 (2018). DOI 10.1088/1748-0221/13/11/C11008. URL https://iopscience.iop. org/article/10.1088/1748-0221/13/11/C11008
- Bednar, J., Petrlik, M., Teixeira Vivaldini, K.C., Saska, M.: Deployment of Reliable Visual Inertial Odometry Approaches for Unmanned Aerial Vehicles in Real-world Environment. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 167–176 (2022). DOI 10.1109/ICUAS54217.2022.9836067. URL https://ieeexplore.ieee.org/document/ 9836067
- Bondyra, A., Gardecki, S., Gasior, P., Giernacki, W.: Performance of Coaxial Propulsion in Design of Multi-rotor UAVs. In: International Conference on Automation, pp. 523–531 (2016). DOI 10. 1007/978-3-319-29357-8_46. URL https://link.springer.com/chapter/10.1007/ 978-3-319-29357-8_46
- Bonilla Licea, D., Ghogho, M., Saska, M.: When Robotics Meets Wireless Communications: An Introductory Tutorial (2022). DOI 10.48550/arXiv.2209.02021. URL https://arxiv.org/ abs/2209.02021. Submitted To IEEE Proceedings
- Bonilla Licea, D., Silano, G., Ghogho, M., Saska, M.: Optimum Trajectory Planning for Multi-Rotor UAV Relays with Tilt and Antenna Orientation Variations. In: European Signal Processing Conference (EUSIPCO), pp. 1586–1590 (2021). DOI 10.23919/EUSIPCO54536.2021.9616232. URL https://ieeexplore.ieee.org/document/9616232
- Bonilla Licea, D., Walter, V., Ghogho, M., Saska, M.: Optical communication-based identification for multi-UAV systems: theory and practice (2022). DOI 10.48550/arXiv.2302.04770. URL https://arxiv.org/abs/2302.04770. Submitted To IEEE Transactions on Robotics
- Brandao, A.S., Smrcka, D., Pairet, E., Nascimento, T., Saska, M.: Side-Pull Maneuver: A Novel Control Strategy for Dragging a Cable-Tethered Load of Unknown Weight Using a UAV. IEEE Robotics and Automation Letters 7(4), 9159–9166 (2022). DOI 10.1109/LRA.2022.3190092. URL https://ieeexplore.ieee.org/document/9827488
- Caballero, A., Silano, G.: A Signal Temporal Logic Motion Planner for Bird Diverter Installation Tasks with Multi-Robot Aerial Systems (2022). DOI 10.48550/arXiv.2210.09750. URL https: //arxiv.org/abs/2210.09750. Submitted To Autonomous Robots
- Calvo, A., Silano, G., Capitan, J.: Mission Planning and Execution in Heterogeneous Teams of Aerial Robots supporting Power Line Inspection Operations. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1644–1649 (2022). DOI 10.1109/ICUAS54217.2022. 9836234. URL https://ieeexplore.ieee.org/document/9836234
- 22. Cataffo, V., Silano, G., Iannelli, L., Puig, V., Glielmo, L.: A Nonlinear Model Predictive Control Strategy for Autonomous Racing of Scale Vehicles. In: 2022 IEEE International Conference on Systems, Man, and Cybernetics, pp. 100–105 (2022). DOI 10.1109/SMC53654.2022.9945279. URL https://ieeexplore.ieee.org/document/9945279
- Chaudhary, A., Nascimento, T., Saska, M.: Controlling a Swarm of Unmanned Aerial Vehicles Using Full-Body k-Nearest Neighbor Based Action Classifier. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 544–551 (2022). DOI 10.1109/ICUAS54217.2022.9836097. URL https://ieeexplore.ieee.org/document/9836097
- 24. Christie, G., Shoemaker, A., Kochersberger, K., Tokekar, P., McLean, L., Leonessa, A.: Radiation search operations using scene understanding with autonomous UAV and UGV. Journal of Field Robotics 34(8), 1450–1468 (2017). DOI 10.1002/rob.21723. URL https://onlinelibrary.wiley.com/doi/10.1002/rob.21723
- Coppola, M., McGuire, K.N., De Wagter, C., de Croon, G.C.H.E.: A Survey on Swarming With Micro Air Vehicles: Fundamental Challenges and Constraints. Frontiers in Robotics and AI 7 (2020). DOI 10.3389/frobt.2020.00018. URL https://www.frontiersin.org/articles/10. 3389/frobt.2020.00018/full
- Demkiv, L., Ruffo, M., Silano, G., Bednar, J., Saska, M.: An Application of Stereo Thermal Vision for Preliminary Inspection of Electrical Power Lines by MAVs. In: Aerial Robotic Systems Physically Interacting with the Environment, pp. 1–8 (2021). DOI 10.1109/AIRPHARO52252.2021. 9571025. URL https://ieeexplore.ieee.org/document/9571025

- Dmytruk, A., Nascimento, T., Ahmad, A., Báča, T., Saska, M.: Safe Tightly-Constrained UAV Swarming in GNSS-denied Environments. In: 2021 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1391–1399 (2021). DOI 10.1109/ICUAS51884.2021.9476794. URL https://ieeexplore.ieee.org/document/9476794
- Dmytruk, A., Silano, G., Bicego, D., Bonilla Licea, D., Saska, M.: A Perception-Aware NMPC for Vision-Based Target Tracking and Collision Avoidance with a Multi-Rotor UAV. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1668–1673 (2022). DOI 10.1109/ICUAS54217.2022.9836071. URL https://ieeexplore.ieee.org/document/ 9836071
- Ebadi, K., Bernreiter, L., Biggie, H., Catt, G., Chang, Y., Chatterjee, A., Denniston, C.E., Deschênes, S.P., Harlow, K., Khattak, S., Nogueira, L., Palieri, M., Petráček, P., Petrlík, M., Reinke, A., Krátký, V., Zhao, S., Agha-mohammadi, A.a., Alexis, K., Heckman, C., Khosoussi, K., Kottege, N., Morrell, B., Hutter, M., Pauling, F., Pomerleau, F., Saska, M., Scherer, S., Siegwart, R., Williams, J.L., Carlone, L.: Present and Future of SLAM in Extreme Underground Environments (2022). DOI 10.48550/arXiv.2208.01787. URL https://arxiv.org/abs/2208.01787. Submitted to IEEE Transactions on Robotics
- Flynn, E.P.: Low-cost approaches to UAV design using advanced manufacturing techniques. In: 2013 IEEE Integrated STEM Education Conference (ISEC), pp. 1–4 (2013). DOI 10.1109/ISECon. 2013.6525199. URL https://ieeexplore.ieee.org/document/6525199
- Foehn, P., Kaufmann, E., Romero, A., Penicka, R., Sun, S., Bauersfeld, L., Laengle, T., Cioffi, G., Song, Y., Loquercio, A., Scaramuzza, D.: Agilicious: Open-source and open-hardware agile quadrotor for vision-based flight. Science Robotics 7(67), eabl6259 (2022). DOI 10.1126/scirobotics.abl6259. URL https://www.science.org/doi/pdf/10.1126/ scirobotics.abl6259
- 32. Giernacki, W., Skwierczyński, M., Witwicki, W., Wroński, P., Kozierski, P.: Crazyflie 2.0 quadrotor as a platform for research and education in robotics and control engineering. In: 2017 22nd International Conference on Methods and Models in Automation and Robotics (MMAR), pp. 37–42 (2017). DOI 10.1109/MMAR.2017.8046794. URL https://ieeexplore.ieee.org/document/ 8046794
- Guo, H., Li, M., Sun, P., Zhao, C., Zuo, W., Li, X.: Lightweight and maintainable rotary-wing UAV frame from configurable design to detailed design. Advances in Mechanical Engineering 13(7) (2021). DOI 10.1177/1687814021103499. URL https://journals.sagepub.com/doi/ full/10.1177/16878140211034999
- 34. Gupta, P.M., Pairet, E., Nascimento, T., Saska, M.: Landing a UAV in Harsh Winds and Turbulent Open Waters. IEEE Robotics and Automation Letters 8(2), 744–751 (2023). DOI 10.1109/LRA. 2022.3231831. URL https://ieeexplore.ieee.org/document/9998066
- Hamandi, M., Usai, F., Sablé, Q., Staub, N., Tognon, M., Franchi, A.: Design of multirotor aerial vehicles: A taxonomy based on input allocation. The International Journal of Robotics Research 40(8-9), 1015–1044 (2021). DOI 10.1177/02783649211025998. URL https://journals. sagepub.com/doi/full/10.1177/02783649211025998
- 36. Hert, D., Baca, T., Petracek, P., Kratky, V., Spurny, V., Petrlik, M., Vrba, M., Zaitlik, D., Stoudek, P., Walter, V., Stepan, P., Horyna, J., Pritzl, V., Silano, G., Bonilla Licea, D., Stibinger, P., Penicka, R., Nascimento, T., Saska, M.: MRS Modular UAV Hardware Platforms for Supporting Research in Real-World Outdoor and Indoor Environments. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1264–1273 (2022). DOI 10.1109/ICUAS54217.2022.9836083. URL https://ieeexplore.ieee.org/document/9836083
- 37. Horyna, J., Baca, T., Walter, V., Albani, D., Hert, D., Ferrante, E., Saska, M.: Decentralized swarms of unmanned aerial vehicles for search and rescue operations without explicit communication. Autonomous Robots pp. 1–17 (2022). DOI 10.1007/s10514-022-10066-5. URL https://link.springer.com/article/10.1007/s10514-022-10066-5
- Horyna, J., Kratky, V., Ferrante, E., Saska, M.: Decentralized multi-robot velocity estimation for UAVs enhancing onboard camera-based velocity measurements. In: 2022 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pp. 11570–11577 (2022). DOI 10.1109/ IROS47612.2022.9981894. URL https://ieeexplore.ieee.org/document/9981894
- 39. Horyna, J., Walter, V., Saska, M.: UVDAR-COM: UV-Based Relative Localization of UAVs with Integrated Optical Communication. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1302–1308 (2022). DOI 10.1109/ICUAS54217.2022.9836151. URL https: //ieeexplore.ieee.org/document/9836151

 Jang, I., Hu, J., Arvin, F., Carrasco, J., Lennox, B.: Omnipotent Virtual Giant for Remote Human–Swarm Interaction. In: 2021 30th IEEE International Conference on Robot & Human Interactive Communication (RO-MAN), pp. 488–494. IEEE (2021). DOI 10.1109/RO-MAN50785. 2021.9515542. URL https://ieeexplore.ieee.org/document/9515542

- 41. Jindal, K., Wang, A., Thakur, D., Zhou, A., V.Spurny, V.Walter, Broughton, G., Krajnik, T., Saska, M., Loianno, G.: Design and Deployment of an Autonomous Unmanned Ground Vehicle for Urban Firefighting Scenarios. Field Robotics 2, 186–202 (2021). DOI 10.55417/fr.2021007. URL https://fieldrobotics.net/Field_Robotics/Volume_1_files/7_Jindal
- 42. Kratky, V., Alcantara, A., Capitan, J., Stepan, P., Saska, M., Ollero, A.: Autonomous Aerial Filming With Distributed Lighting by a Team of Unmanned Aerial Vehicles. IEEE Robotics and Automation Letters 6(4), 7580–7587 (2021). DOI 10.1109/LRA.2021.3098811. URL https: //ieeexplore.ieee.org/document/9495218
- Kratky, V., Petracek, P., Baca, T., Saska, M.: An Autonomous Unmanned Aerial Vehicle System for Fast Exploration of Large Complex Indoor Environments. Journal of Field Robotics 38(8) (2021). DOI 10.1002/rob.22021. URL https://onlinelibrary.wiley.com/doi/abs/ 10.1002/rob.22021
- 44. Kratky, V., Petracek, P., Nascimento, T., Cadilova, M., Skobrtal, M., Stoudek, P., Saska, M.: Safe Documentation of Historical Monuments by an Autonomous Unmanned Aerial Vehicle. ISPRS International Journal of Geo-Information 10(11) (2021). DOI/10.3390/ijgi10110738. URL https: //www.mdpi.com/2220-9964/10/11/738/htm
- Kratky, V., Petracek, P., Spurny, V., Saska, M.: Autonomous Reflectance Transformation Imaging by a Team of Unmanned Aerial Vehicles. IEEE Robotics and Automation Letters 5(2), 2302– 2309 (2020). DOI 10.1109/LRA.2020.2970646. URL https://ieeexplore.ieee.org/ document/8977312
- 46. Krizek, M., Horyna, J., Saska, M.: Swarming of Unmanned Aerial Vehicles by Sharing Distributed Observations of Workspace. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 300–309 (2022). DOI 10.1109/ICUAS54217.2022.9836073. URL https: //ieeexplore.ieee.org/document/9836073
- 47. Loianno, G., Brunner, C., McGrath, G., Kumar, V.: Estimation, Control, and Planning for Aggressive Flight With a Small Quadrotor With a Single Camera and IMU. IEEE Robotics and Automation Letters 2(2), 404–411 (2017). DOI 10.1109/LRA.2016.2633290. URL https://ieeexplore. ieee.org/document/7762111
- Loianno, G., Spurny, V., Thomas, J., Baca, T., Thakur, D., Hert, D., Penicka, R., Krajnik, T., Zhou, A., Cho, A., Saska, M., Kumar, V.: Localization, Grasping, and Transportation of Magnetic Objects by a team of MAVs in Challenging Desert like Environments. IEEE Robotics and Automation Letters 3(3), 1576–1583 (2018). DOI 10.1109/LRA.2018.2800121. URL https: //ieeexplore.ieee.org/document/8276269
- Ma, J., Lai, E.M.K., Ren, J.: On the Timing of Operator Commands for the Navigation of a Robot Swarm. In: 2018 15th International Conference on Control, Automation, Robotics and Vision (ICARCV), pp. 1634–1639 (2018). DOI 10.1109/ICARCV.2018.8581236. URL https: //ieeexplore.ieee.org/document/8581236
- Martin, P.G., Kwong, S., Smith, N., Yamashiki, Y., Payton, O.D., Russell-Pavier, F., Fardoulis, J.S., Richards, D., Scott, T.B.: 3D unmanned aerial vehicle radiation mapping for assessing contaminant distribution and mobility. International Journal of Applied Earth Observation and Geoinformation 52, 12–19 (2016). DOI 10.1016/j.jag.2016.05.007. URL https://www.sciencedirect. com/science/article/pii/S0303243416300733
- Meier, L., Honegger, D., Pollefeys, M.: PX4: A node-based multithreaded open source robotics framework for deeply embedded platforms. In: 2015 IEEE International Conference on Robotics and Automation (ICRA), pp. 6235–6240 (2015). DOI 10.1109/ICRA.2015.7140074. URL https: //ieeexplore.ieee.org/document/7140074
- 52. Mohta, K., Watterson, M., Mulgaonkar, Y., Liu, S., Qu, C., Makineni, A., Saulnier, K., Sun, K., Zhu, A., Delmerico, J., Karydis, K., Atanasov, N., Loianno, G., Scaramuzza, D., Daniilidis, K., Taylor, C.J., Kumar, V.: Fast, autonomous flight in gps-denied and cluttered environments. Journal of Field Robotics 35(1), 101–120 (2018). DOI https://doi.org/10.1002/rob.21774. URL https: //onlinelibrary.wiley.com/doi/abs/10.1002/rob.21774
- Musil, T., Petrlik, M., Saska, M.: SphereMap: Dynamic Multi-Layer Graph Structure for Rapid Safety-Aware UAV Planning. IEEE Robotics and Automation Letters 7(4), 11007–11014 (2022). DOI 10.1109/LRA.2022.3195194. URL https://ieeexplore.ieee.org/document/ 9847042

- 54. Nascimento, T.P., Saska, M.: Position and attitude control of multi-rotor aerial vehicles: A survey. Annual Reviews in Control 48, 129–146 (2019). DOI 10.1016/j.arcontrol. 2019.08.004. URL https://www.sciencedirect.com/science/article/pii/ S1367578819300483
- Nekovar, F., Faigl, J., Saska, M.: Multi-Tour Set Traveling Salesman Problem in Planning Power Transmission Line Inspection. IEEE Robotics and Automation Letters 6(4), 6196–6203 (2021). DOI 10.1109/LRA.2021.3091695. URL https://ieeexplore.ieee.org/document/ 9463765
- 56. Nekovar, F., Faigl, J., Saska, M.: Multi-vehicle Dynamic Water Surface Monitoring (2023). DOI 10.48550/arXiv.2302.11991. URL https://arxiv.org/abs/2302.11991. Submitted To IEEE Robotics and Automation Letters
- Nekovář, F., Faigl, J., Saska, M.: Vehicle Fault-Tolerant Robust Power Transmission Line Inspection Planning. In: 2022 IEEE 27th International Conference on Emerging Technologies and Factory Automation (ETFA), pp. 1–4 (2022). DOI 10.1109/ETFA52439.2022.9921692. URL https: //ieeexplore.ieee.org/document/9921692
- Novak, F., Walter, V., Petracek, P., Baca, T., Saska, M.: Fast collective evasion in self-localized swarms of unmanned aerial vehicles. Bioinspiration & biomimetics 16(6), 066025 (2021). DOI 10.1088/1748-3190/ac3060. URL https://iopscience.iop.org/article/10.1088/ 1748-3190/ac3060
- 59. Orekhov, V., Chung, T.: The darpa subterranean challenge: A synopsis of the circuits stage. Field Robotics 2, 735-747 (2022). DOI 10.55417/fr.2022024. URL https://www. journalfieldrobotics.org/Field_Robotics/Volume_2_files/Vol2_24.pdf
- Papaioannidis, C., Makrygiannis, D., Mademlis, I., Pitas, I.: Learning Fast and Robust Gesture Recognition. In: 2021 29th European Signal Processing Conference (EUSIPCO), pp. 761–765 (2021). DOI 10.23919/EUSIPCO54536.2021.9616227. URL https://ieeexplore.ieee. org/document/9616227
- Patel, V.V., Liarokapis, M.V., Dollar, A.M.: Open Robot Hardware: Progress, Benefits, Challenges, and Best Practices. IEEE Robotics & Automation Magazine pp. 2–29 (2022). DOI 10.1109/MRA. 2022.3225725. URL https://ieeexplore.ieee.org/document/9992182
- 62. Perez, D., Alcantara, A., Capitan, J.: Distributed Trajectory Planning for a Formation of Aerial Vehicles Inspecting Wind Turbines. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 646–654 (2022). DOI 10.1109/ICUAS54217.2022.9836173. URL https://ieeexplore.ieee.org/document/9836173
- Petracek, P., Kratky, V., Baca, T., Petrlik, M., Saska, M.: New Era in Cultural Heritage Preservation: Cooperative Aerial Autonomy: Supervised Autonomy for Fast Digitalization of Difficult-to-Access Interiors of Historical Monuments. IEEE Robotics & Automation Magazine pp. 2–19 (2023). DOI 10.1109/MRA.2023.3244423. URL https://ieeexplore.ieee.org/document/ 10056379
- Petracek, P., Kratky, V., Petrlik, M., Baca, T., Kratochvil, R., Saska, M.: Large-Scale Exploration of Cave Environments by Unmanned Aerial Vehicles. IEEE Robotics and Automation Letters 6(4), 7596–7603 (2021). DOI 10.1109/LRA.2021.3098304. URL https://ieeexplore.ieee. org/document/9492802
- Petracek, P., Kratky, V., Saska, M.: Dronument: System for Reliable Deployment of Micro Aerial Vehicles in Dark Areas of Large Historical Monuments. IEEE Robotics and Automation Letters 5(2), 2078–2085 (2020). DOI 10.1109/LRA.2020.2969935. URL https://ieeexplore.ieee. org/document/8972370
- Petracek, P., Walter, V., Baca, T., Saska, M.: Bio-Inspired Compact Swarms of Micro Aerial Vehicles without Communication and External Localization. Bioinspiration & Biomimetics 16(2) (2020). DOI 10.1088/1748-3190/abc6b3. URL https://iopscience.iop.org/article/10.1088/1748-3190/abc6b3
- Petrlik, M., Baca, T., Hert, D., Vrba, M., Krajnik, T., Saska, M.: A Robust UAV System for Operations in a Constrained Environment. IEEE Robotics and Automation Letters 5 (2020). DOI 10.1109/ LRA.2020.2970980. URL https://ieeexplore.ieee.org/document/8979150
- Petrlik, M., Petracek, P., Kratky, V., Musil, T., Stasinchuk, Y., Vrba, M., Baca, T., Hert, D., Pecka, M., Svoboda, T., Saska, M.: UAVs Beneath the Surface: Cooperative Autonomy for Subterranean Search and Rescue in DARPA SubT. Field Robotics 3, 1–68 (2023). DOI 10.55417/fr. 2023001. URL https://fieldrobotics.net/Field_Robotics/Volume_3_files/ Vol3_01.pdf

 Pritzl, V., Stepan, P., Saska, M.: Autonomous Flying into Buildings in a Firefighting Scenario. In: 2021 IEEE International Conference on Robotics and Automation (ICRA), pp. 239–245. IEEE (2021). DOI 10.1109/ICRA48506.2021.9560789. URL https://ieeexplore.ieee.org/ document/9560789

- Pritzl, V., Vrba, M., Stepan, P., Saska, M.: Cooperative Navigation and Guidance of a Micro-Scale Aerial Vehicle by an Accompanying UAV using 3D LiDAR Relative Localization. In: 2022 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 526–535 (2022). DOI 10.1109/ICUAS54217.2022.9836116. URL https://ieeexplore.ieee.org/document/ 9836116
- Pritzl, V., Vrba, M., Tortorici, C., Ashour, R., Saska, M.: Adaptive estimation of UAV altitude in complex indoor environments using degraded and time-delayed measurements with time-varying uncertainties. Robotics and Autonomous Systems 160, 104315 (2023). DOI 10.1016/j.robot.2022.104315. URL https://www.sciencedirect.com/science/article/pii/S0921889022002044
- 72. Ren, Z., Meng, J., Yuan, J.: Depth camera based hand gesture recognition and its applications in human-computer-interaction. In: 2011 8th International Conference on Information, Communications & Signal Processing (ICICS), pp. 1–5 (2011). DOI 10.1109/ICICS.2011.6173545. URL https://ieeexplore.ieee.org/document/6173545
- 73. Rodriguez, F., Diaz-Banez, J.M., Sanchez-Laulhe, E., Capitan, J., Ollero, A.: Kinodynamic planning for an energy-efficient autonomous ornithopter. Computers & Industrial Engineering 163, 107814 (2021). DOI 10.1016/j.cie.2021.107814. URL https://www.sciencedirect.com/science/article/pii/S036083522100718X
- 74. Roucek, T., Pecka, M., Cizek, P., Petricek, T., Bayer, J., Salansky, V., Azayev, T., Hert, D., Petrlik, M., Baca, T., Spurny, V., Kratky, V., Petracek, P., Baril, D., Vaidis, M., Kubelka, V., Pomerleau, F., Faigl, J., Zimmermann, K., Saska, M., Svoboda, T., Krajnik, T.: System for multi-robotic exploration of underground environments CTU-CRAS-NORLAB in the DARPA Subterranean Challenge. Field Robotics 2, 1779–1818 (2022). DOI 10.55417/fr.2022055. URL https://fieldrobotics.net/Field_Robotics/Volume_2_files/Vol2_55.pdf
- 75. Sa, I., Kamel, M., Burri, M., Bloesch, M., Khanna, R., Popović, M., Nieto, J., Siegwart, R.: Build your own visual-inertial drone: A cost-effective and open-source autonomous drone. IEEE Robotics & Automation Magazine 25(1), 89–103 (2018). DOI 10.1109/MRA.2017.2771326. URL https: //ieeexplore.ieee.org/document/8233182
- 76. Saska, M., Kratky, V., Spurny, V., Baca, T.: Documentation of dark areas of large historical buildings by a formation of unmanned aerial vehicles using model predictive control. In: IEEE 2019 24th IEEE International Conference on Emerging Technologies and Factory Automation (ETFA), pp. 1–8 (2017). DOI 10.1109/ETFA.2017.8247654. URL https://ieeexplore.ieee.org/ abstract/document/8247654
- 77. Schacht-Rodriguez, R., Ortiz-Torres, G., Garcia-Beltran, C.D., Astorga-Zaragoza, C.M., Ponsart, J.C., Perez-Estrada, A.J.: Design and development of a uav experimental platform. IEEE Latin America Transactions 16(5), 1320–1327 (2018). DOI 10.1109/TLA.2018.8408423. URL https://ieeexplore.ieee.org/document/8408423
- Silano, G., Aucone, E., Iannelli, L.: CrazyS: A Software-In-The-Loop Platform for the Crazyfie 2.0 Nano-Quadcopter. In: 2018 26th Mediterranean Conference on Control and Automation (MED), pp. 1–6 (2018). DOI 10.1109/MED.2018.8442759. URL https://ieeexplore.ieee.org/ document/8442759
- 79. Silano, G., Baca, T., Penicka, R., Liuzza, D., Saska, M.: Power Line Inspection Tasks With Multi-Aerial Robot Systems Via Signal Temporal Logic Specifications. IEEE Robotics and Automation Letters 6(2), 4169–4176 (2021). DOI 10.1109/LRA.2021.3068114. URL https: //ieeexplore.ieee.org/document/9384182
- Silano, G., Bednar, J., Nascimento, T., Capitan, J., Saska, M., Ollero, A.: A Multi-Layer Software Architecture for Aerial Cognitive Multi-Robot Systems in Power Line Inspection Tasks. In: 2021 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1624–1629 (2021). DOI 10.1109/ICUAS51884.2021.9476813. URL https://ieeexplore.ieee.org/document/ 9476813
- Silano, G., Iannelli, L.: Robot Operating System (ROS): The Complete Reference (Volume 4), chap. CrazyS: a software-in-the-loop simulation platform for the Crazyflie 2.0 nano-quadcopter, pp. 81– 115. Springer International Publishing, Cham (2020). DOI 10.1007/978-3-030-20190-6_4. URL https://link.springer.com/chapter/10.1007/978-3-030-20190-6_4

- Smrcka, D., Baca, T., Nascimento, T., Saska, M.: Admittance Force-Based UAV-Wall Stabilization and Press Exertion for Documentation and Inspection of Historical Buildings. In: 2021 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 552–559 (2021). DOI 10.1109/ICUAS51884.2021.9476873. URL https://ieeexplore.ieee.org/document/9476873
- Spurny, V., Baca, T., Saska, M., Penicka, R., Krajnik, T., et al.: Cooperative Autonomous Search, Grasping and Delivering in a Treasure Hunt Scenario by a Team of UAVs. Journal of Field Robotics 36(1), 125–148 (2019). DOI 10.1002/rob.21816. URL https://onlinelibrary.wiley. com/doi/abs/10.1002/rob.21816
- 84. Spurny, V., Pritzl, V., Walter, V., Petrlik, M., Baca, T., Stepan, P., Zaitlik, D., Saska, M.: Autonomous Firefighting Inside Buildings by an Unmanned Aerial Vehicle. IEEE Access 9, 15872–15890 (2021). DOI 10.1109/ACCESS.2021.3052967. URL https://ieeexplore.ieee.org/document/9328798
- 85. Stasinchuk, Y., Vrba, M., Baca, T., Spurny, V., Petrlik, M., Hert, D., D.Zaitlik, M.Saska: A Multi-MAV System For Autonomous Multiple Targets Elimination in the MBZIRC 2020 competition. Field Robotics 2, 1697–1720 (2021). DOI 10.55417/fr.2022052. URL http:// fieldrobotics.net/Field_Robotics/Volume_2_files/Vol2_52
- 86. Stasinchuk, Y., Vrba, M., Petrlik, M., Baca, T., Spurny, V., Hert, D., Zaitlik, D., Nascimento, T., Saska, M.: A Multi-UAV System for Detection and Elimination of Multiple Targets. In: 2021 IEEE International Conference on Robotics and Automation (ICRA), pp. 555–561 (2021). DOI 10.1109/ICRA48506.2021.9562057. URL https://ieeexplore.ieee.org/document/ 9562057
- 87. Stepan, P., Krajnik, T., Petrlik, M., Saska, M.: Vision techniques for on-board detection, following and mapping of moving targets. Journal of Field Robotics 36(1), 252–269 (2019). DOI 10.1002/rob. 21850. URL https://onlinelibrary.wiley.com/doi/10.1002/rob.21850
- Stibinger, P., Baca, T., Saska, M.: Localization of Ionizing Radiation Sources by Cooperating Micro Aerial Vehicles With Pixel Detectors in Real-Time. IEEE Robotics and Automation Letters 5, 3634– 3641 (2020). DOI 10.1109/LRA.2020.2978456. URL https://ieeexplore.ieee.org/ document/9024023
- Stibinger, P., Broughton, G., Majer, F., Rozsypalek, Z., Wang, A., Jindal, K., Zhou, A., Thakur, D., Loianno, G., Krajnik, T., Saska, M.: Towards new frontiers in mobile manipulation: Team CTU-UPenn-NYU at MBZIRC 2020. Field Robotics 2, 75–106 (2022). DOI 10.55417/fr. 2022004. URL https://fieldrobotics.net/Field_Robotics/Volume_2_files/ Vol2_04.pdf
- 90. Stingu, E., Lewis, F.: A Hardware Platform for Research in Helicopter UAV Control. Unmanned Aircraft Systems 54, 387–406 (2009). DOI 10.1007/978-1-4020-9137-7_21. URL https:// link.springer.com/chapter/10.1007/978-1-4020-9137-7_21
- Towler, J., Krawiec, B., Kochersberger, K.: Radiation mapping in post-disaster environments using an autonomous helicopter. Remote Sensing 4(7), 1995–2015 (2012). DOI 10.3390/rs4071995. URL https://www.mdpi.com/2072-4292/4/7/1995
- 92. Turecek, D., Jakubek, J., Trojanova, E., Sefc, L.: Single layer compton camera based on timepix3 technology. Journal of Instrumentation 15(01) (2020). DOI 10.1088/1748-0221/15/01/ C01014. URL https://iopscience.iop.org/article/10.1088/1748-0221/15/ 01/C01014
- 93. Vrba, M., Hert, D., Saska, M.: Onboard Marker-Less Detection and Localization of Non-Cooperating Drones for Their Safe Interception by an Autonomous Aerial System. IEEE Robotics and Automation Letters 4(4), 3402–3409 (2019). DOI 10.1109/LRA.2019.2927130. URL https://ieeexplore.ieee.org/document/8756100
- Vrba, M., Saska, M.: Marker-Less Micro Aerial Vehicle Detection and Localization Using Convolutional Neural Networks. IEEE Robotics and Automation Letters 5(2), 2459–2466 (2020). DOI 10. 1109/LRA.2020.2972819. URL https://ieeexplore.ieee.org/document/8988144
- 95. Vrba, M., Stasinchuk, Y., Báča, T., Spurný, V., Petrlík, M., Heřt, D., Žaitlík, D., Saska, M.: Autonomous capture of agile flying objects using UAVs: The MBZIRC 2020 challenge. Robotics and Autonomous Systems 149, 103970 (2022). DOI https://doi.org/10.1016/j.robot. 2021.103970. URL https://www.sciencedirect.com/science/article/pii/ S0921889021002396
- 96. Vrba, M., Walter, V., Saska, M.: On Onboard LiDAR-based Flying Object Detection (2023). DOI 10.48550/arXiv.2303.05404. URL https://arxiv.org/abs/2303.05404. Submitted To IEEE Transactions on Robotics

97. Štibinger, P., Broughton, G., Majer, F., Rozsypálek, Z., Wang, A., Jindal, K., Zhou, A., Thakur, D., Loianno, G., Krajník, T., Saska, M.: Mobile Manipulator for Autonomous Localization, Grasping and Precise Placement of Construction Material in a Semi-structured Environment. IEEE Robotics and Automation Letters 6(2), 2595–2602 (2021). DOI 10.1109/LRA.2021.3061377. URL https://ieeexplore.ieee.org/document/9361167

- Walter, V., Saska, M., Franchi, A.: Fast mutual relative localization of UAVs using ultraviolet LED markers. In: 2018 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1217– 1226 (2018). DOI 10.1109/ICUAS.2018.8453331. URL https://ieeexplore.ieee.org/ document/8453331
- 99. Walter, V., Spurny, V., Petrlik, M., Baca, T., Zaitlik, D., Demkiv, L., Saska, M.: Extinguishing Real Fires by Fully Autonomous Multirotor UAVs in the MBZIRC 2020 Competition. Field Robotics 2, 406–436 (2021). DOI 10.55417/fr.2022015. URL https://fieldrobotics.net/Field_ Robotics/Volume_2_files/Vol2_15
- 100. Walter, V., Spurny, V., Petrlik, M., Baca, T., Zaitlik, D., Saska, M.: Extinguishing of Ground Fires by Fully Autonomous UAVs Motivated by the MBZIRC 2020 Competition. In: 2021 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 787–793 (2021). DOI 10.1109/ICUAS51884. 2021.9476723. URL https://ieeexplore.ieee.org/document/9476723
- 101. Walter, V., Staub, N., Saska, M., Franchi, A.: Mutual Localization of UAVs based on Blinking Ultraviolet Markers and 3D Time-Position Hough Transform. In: 2018 IEEE 14th International Conference on Automation Science and Engineering (CASE), pp. 298–303 (2018). DOI 10.1109/COASE. 2018.8560384. URL https://ieeexplore.ieee.org/document/8560384
- 102. Walter, V., Vrba, M., Saska, M.: On training datasets for machine learning-based visual relative localization of micro-scale UAVs. In: 2020 IEEE International Conference on Robotics and Automation (ICRA), pp. 10674–10680 (2020). DOI 10.1109/ICRA40945.2020.9196947. URL https://ieeexplore.ieee.org/document/9196947
- 103. Walter, W., Staub, N., Franchi, A., Saska, M.: UVDAR System for Visual Relative Localization With Application to Leader-Follower Formations of Multirotor UAVs. IEEE Robotics and Automation Letters 4(3), 2637–2644 (2019). DOI 10.1109/LRA.2019.2901683. URL https: //ieeexplore.ieee.org/document/8651535
- 104. Wang, S., Njau, C.E., Jiang, Z.: Design and Implementation of Multi-UAV Cooperation Search Experimental platform. In: 2021 5th International Conference on Robotics and Automation Sciences (ICRAS), pp. 94–98 (2021). DOI 10.1109/ICRAS52289.2021.9476222. URL https: //ieeexplore.ieee.org/document/9476222
- 105. Wu, M., Zhu, X., Ma, L., Wang, J., Bao, W., Li, W., Fan, Z.: Torch: Strategy evolution in swarm robots using heterogeneous-homogeneous coevolution method. Journal of Industrial Information Integration 25, 100239 (2022). DOI 10.1016/j.jii.2021.100239. URL https://www. sciencedirect.com/science/article/pii/S2452414X2100039X